Aalborg Universitet



Data Mining Applications to Fault Diagnosis in Power Electronic Systems

A Systematic Review

Moradzadeh, Arash; Mohammadi-Ivatloo, Behnam; Pourhossein, Kazem; Anvari-Moghaddam, Amjad

Published in: I E E E Transactions on Power Electronics

DOI (link to publication from Publisher): 10.1109/TPEL.2021.3131293

Publication date: 2022

Document Version Accepted author manuscript, peer reviewed version

Link to publication from Aalborg University

Citation for published version (APA):

Moradzadeh, A., Mohammadi-Ivatlóo, B., Pourhossein , K., & Anvari-Moghaddam, A. (2022). Data Mining Applications to Fault Diagnosis in Power Electronic Systems: A Systematic Review. *I E E E Transactions on Power Electronics*, 37(5), 6026-6050. https://doi.org/10.1109/TPEL.2021.3131293

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
- You may not further distribute the material or use it for any profit-making activity or commercial gain You may freely distribute the URL identifying the publication in the public portal -

Take down policy

If you believe that this document breaches copyright please contact us at vbn@aub.aau.dk providing details, and we will remove access to the work immediately and investigate your claim.

Data Mining Applications to Fault Diagnosis in Power Electronic Systems: A Systematic Review

Arash Moradzadeh, Student Member, IEEE, Behnam Mohammadi-Ivatloo, Senior Member, IEEE, Kazem Pourhossein, Amjad Anvari-Moghaddam, Senior Member, IEEE

Abstract— Early fault detection in power electronic systems (PESs) to maintain reliability is one of the most important issues that has been significantly addressed in recent years. In this paper, after reviewing various literature based on fault detection in PESs, data mining-based techniques including artificial neural network, machine learning, and deep learning algorithms are introduced. Then, the fault detection routine in PESs is expressed by introducing signal measurement sensors and how to extract the feature from it. Finally, based on studies, the performance of various data mining methods in detecting PESs faults is evaluated. The results of evaluations show that the deep learning-based techniques given the ability of feature extraction from measured signals are significantly more effective than other methods and as an ideal tool for future applications in power electronics industry are introduced.

Index Terms—Power electronic systems, reliability, fault detection, fault tolerant, artificial neural network, machine learning, deep learning.

I. INTRODUCTION

NOWADAYS, electrical energy has become an influential factor in the scientific, economic and welfare fields of human daily life. In recent years, the expansion of electrical energy applications and the increase of electrical energy consumers have made distributed generation (DGs) dramatically replace traditional power systems [1], [2]. On the other hand, DGs such as renewable energy sources (RESs) and energy storage systems have been widely used to reduce fossil fuel consumption and solve environmental problems. But the important point is that the production, storage and utilization of electrical energy in the economic and daily life cycle require power electronic systems (PESs) [3]. PESs have a significant role in integrating RESs, energy management, and reliability of power grids, and other related infrastructures and systems [4], [5]. Energy/power conversion using PESs is easy and low cost. Despite all the advantages of PESs, their high vulnerability to natural disasters, frequent switching in harsh environment, and etc. that results in power outages or system shutdown and accordingly increased cost of operating, is one of their biggest disadvantages. Long-term sustainability without power interruption is one of the most important factors in the needs of PESs applications. In most cases, severe environmental conditions such as high temperatures, over voltage and over current, wear-out of electrical components, radiation, vibration and mechanical damages, thermal damages, hardware design or control defects, and electromagnetic stresses are major causes of critical failures in PESs. Some studies have shown that semiconductors of the primary side (low voltage, high current) and resonance elements used in PESs can be the main sources of damage due to various factors. Faults that occur mainly in different parts of the PESs are divided into two categories of structural faults (hard faults) and parametric faults (soft faults) [6]-[8]. Each of the hard and soft faults are divided into various types of anomalies, which are introduced as follow:

A. Hard faults

Hard faults occur due to drastic changes in the value of parameters related to the components or circuit structures in the PESs. These faults are observed in two cases of SC fault and OC fault. Hard faults can have effects such as a sudden increase in current and a sudden voltage drop in PESs. Thus, the occurrence of a hard fault provides the basis for serious and catastrophic damage to the entire system. Hard faults generally do not occur directly in the system and are happen often due to the intensity and persistence of soft faults in the circuit [9], [10].

B. Soft faults

Soft faults mainly refer to the parameters of the circuit components from their tolerance range, but they do not affect the circuit connections. Soft faults are known as parameter drift and cause a gradual decrease in system performance and ultimately cause aging and wearing out [11]. The occurrence of

Arash Moradzadeh is with the Faculty of Electrical and Computer Engineering, University of Tabriz, Tabriz 51666, Iran, (email: arash.moradzadeh@tabrizu.ac.ir; arash.moradzadeh@ieee.org).

Behnam Mohammadi-Ivatloo is with the Faculty of Electrical and Computer Engineering, University of Tabriz, Tabriz 51666, Iran (e-mail: mohammadi@ieee.org).

Kazem Pourhossein is with the Department of Electrical Engineering, Tabriz Branch, Islamic Azad University, Tabriz, Iran (e-mail: k.pourhossein@iaut.ac.ir).

Amjad Anvari-Moghaddam is with the Department of Energy (AAU Energy), Aalborg University, 9220 Aalborg, Denmark (aam@energy.aau.dk).

soft faults does not completely interrupt the operation of the circuit in PESs, but causes an unacceptable operation of the circuit by creating an unwanted output. Studies and experiments have shown that the soft faults can become a hard fault if they are not detected and fixed in a timely manner [9].

In recent years, various studies have been conducted on the stability and reliability of power electronic converters, and based on more than 200 products from 80 companies, it has been concluded that capacitors and power semiconductor devices include more than 50% of PES failures (As shown in Fig. 1). Occurrence of any faults in these systems will cause serious damages to the entire system.

Maintenance in PESs is an topic that has posed many challenges and includes reliability, stability, condition monitoring, fault detection, and useful life estimation [12]. Several review papers over the past decade have addressed this issue [10], [13]–[15]. Advanced analysis of condition monitoring and fault diagnosis in PESs is reviewed in [13].

However, the study includes very limited methods of faultdiagnosis based on Artificial Neural Network (ANN) algorithms. Authors of [10] examine the condition monitoring techniques of capacitors in power electronic converters, which also emphasize the methods of parameter identification based on ANNs. In addition, a variety of ANN-based techniques called the dynamic Bayesian network and object-oriented Bayesian network have been employed in [16], [17], for industrial applications such as transient and intermittent fault detection in complex electronic systems. In another valuable study [18], the hybrid applications rule-based algorithm and back propagation neural network (BPNN) for fault detection in a diesel engine are introduced. In this study, the signals are processed via wavelet threshold denoising and ensemble empirical mode decomposition. Early fault detection in permanent magnet synchronous motor has been done in [19] by presenting data-based approaches called Bayesian network. In this study, to improve the fault detection process, wavelet threshold denoising and minimum entropy deconvolution techniques are employed to pre-processing and denoising the input signals. In [14], a summary of machine learning methods used to manage the reliability of energy systems has been provided. Another valuable study [15], examines the application of ANN



Fig. 1. Percentage of PESs failures [7]

techniques in PESs. This paper is also generally limited to ANN algorithms.

Accurate and early detection of any of the faults in PESs is one of the most important issues that has created many challenges for researchers and craftsmen in the fields of power electronics and industrial electronics. Most of the recent research has identified, analyzed, and diagnosed all types of faults in PESs in different ways. Some of them have succeeded, but some others have failed to detect the fault correctly. Identifying and diagnosis any faults in PESs requires an evaluation of the impact of each fault on the system [4], [20]. So far, in various studies several categories of fault detection methods have been introduced and employed in PESs. Frequency-domain based techniques, Wavelet Transform (WT), ANN algorithms, machine learning-based procedures, and deep learning applications are the most important methods discussed in identifying faults of PESs. Fault detection in PESs began with conventional methods and the use of numerous sensors. These methods were expensive and suffered from problems in timely and accurate fault detection. Meanwhile, with the advent of the Internet of Things, the use of intelligent sensors has led to the exchange of a wide range of data in energy systems technologies and PES applications. After that, traditional and conventional methods did not perform reasonably well in the face of high volumes data. Thus, the increasing volume of data provided a good basis for the application of data mining science and other techniques such as ANN, machine learning, and deep learning. In addition, the challenges and specific features of PESs such as high sensitivity in condition monitoring for aging detection, the need for online monitoring, and high adjustment speed in control, has increased the need for data mining applications in PESs dramatically.

In this paper, the application of data mining techniques such as ANN, machine learning, and deep learning algorithms in detection of PES faults are reviewed in detail. By examining the machine learning and deep learning techniques, the gaps in previous research that mainly focus on ANN techniques are filled. Furthermore, this paper introduces the types of faults in PESs and the necessity of timely fault detection, investigates the impact of each fault on the system, and reviews a variety of fault diagnosis methods. Finally, the fault prognosis models that can be a very important step in the increasing the reliability of PESs will be explored.

The rest of the paper is organized as follows. Section II reviews studies conducted to identify faults in PESs. Section III introduces and categorize the fault diagnosis methods in PESs. Routine of fault diagnosis in PESs is presented in Section IV. Section V introduces the sensors and instruments for signal measurement in PESs. Section VI describes fault-tolerant in PESs. Finally, the concluding remarks are presented in Section VII.

II. HISTORICAL AND LITERATURE REVIEW OF FAULT DETECTION IN PESS

The main objective of this study was to review and evaluate the performance of each of the methods used to detect faults in PESs. This evaluation includes all studies conducted on fault detection from the beginning to the present and looks at future challenges over time. Thus, the initial studies related to fault detection in PESs in the form of a sub-section called historical review and other recent studies under the title of the literature review are presented in the continuation of this section.

A. Historical review of fault detection in PESs

So far, many solutions have been utilized based on the classification presented in the previous section to detect faults related to PESs. A probabilistic applications of ANNs called radial basis function (RBF) was reported in 2003 in [21], to identify SC fault associated with an inverter driver. In another valuable study [22], using ANN algorithms, single commutation failure, double not successive commutation failure, and double successive commutation failure in a converter used in high-voltage direct current (HVDC) system have been identified. In [23] and [24], the ANN applications are employed to detect OC faults and hard faults (which refer to the analogue part of the circuit) in a delta-sigma converter, respectively. Transistor switch faults in a voltage source inverter have been identified in [25] by a new model of ANN based on the controller of space vector modulation.

As can be seen from the reviewed literature, the ANN applications have been used continuously for several years as fault diagnostic methods in PESs. In 2007, two advanced neural network called self-organizing maps (SOM) and learning vector quantization have been introduced and used to detect the SC and overcurrent faults of transistor utilized in isolated DC-DC converters in multi-phase multilevel motor drives [26]. Later, a hierarchical fuzzy-based diagnostic solution was introduced in [27] to identify the SC and AC faults of switches in a direct current (DC)-motor-based brake-by-wire system. In a PV system, fault in DC transmission line, commutation fault in one of the thyristors of the inverter, and single-phase fault in the AC system in the inverter side have been diagnosed using an improved ANN algorithm called, ADAptive linear neuron [28]. The SC fault associated with converter-fed induction motors is identified in [29] using the Feed Forward Neural Network method and based on the FFT signals of the system.

B. Literature review of Fault detection in PESs

With the advancement of technology and the expansion of the data volume related to various issues of power systems and PESs, the performance of ANNs was reduced to some extent. To improve the detection process and increase the accuracy of detection operations, in 2009, a supervised kernel-based method and support vector machine (SVM) were first proposed as applications of machine learning in power electronics [30]. In the study, the proposed methods were employed to detect OC fault of switch in pulse-width modulation (PWM) voltage fed power converter of brushless DC motor drive. In another valuable work [31], rotor faults corresponding to a converterfed induction motor and changeable rotors have been categorized using machine learning techniques called radial basis neural network and k-means. The SVM technique has been proposed in [32] to identify thyristors faults in a threephase full-bridge controlled rectifier.

The proper performance of machine learning techniques in various studies led to the rapid development of the use of these methods in the science and industry related to power electronics. For the first time in 2013, a new machine learning technique called Extreme Learning Machine (ELM) has been proposed to predict the inter-turn SC fault in a three-phase converter-fed induction motor [33]. In the same study, in order to express the effectiveness of the suggested procedure, one of the ANN methods called MLP is also utilized to identify faults, which the results emphasize the superiority of the ELM method. Identification and classification of six types of faults including the SC, pulse loss, AC single-phase grounding, AC two-phase grounding, AC three-phase grounding, and DC grounding in an HVDC converter have been performed via the SVM in [34].

In the process of using different machine learning techniques, a new algorithm called Least-Squares Support-Vector Machine (LSSVM) in 2014 has been introduced and utilized to detect the SC fault in a three-phase squirrel-cage induction motor fed by a sinusoidal PWM converter [35]. In the same work, the effectiveness and high performance of the suggested method has been compared to other ANN and machine learning techniques such as MLP, ELM, SVM, and the Minimal Learning Machine was approved. Various faults in Inverter side of a 12-pulse Line HVDC commutated converter include Single Line to Ground (Rectifier side), Double Line to Ground (Rectifier side), Line-to-Line (Rectifier side), and DC Fault have been identified and categorized by an ANN technique called Levenberg Marquardt backpropagation algorithm in [36].

As stated in the literature, the utilize of data mining techniques in detecting PES faults began with the employing supervised algorithms and over time led to a significant increase in their use. With the expansion of data volume, these methods encountered some problems in the training phase, such as overfitting or data missing. In 2015, in two valuable studies [37], [38], one of the unsupervised methods of data mining called Principle Component Analysis (PCA) has been introduced for the first time to fault detection and improve the problems of supervised methods. In those studies, the PCA technique for detecting the SC switch fault in a cascaded Hbridge multilevel (5-level) inverter has been combined with the FFT and SVM methods to significantly increase the fault detection accuracy. In [39], Discrete Wavelet Transform (DWT) and Fuzzy Inference Logic methods have been selected to detect the various faults include DC SC to ground, OC of insulated gate bipolar transistor (IGBT), SC damage of IGBT, DC link capacitor, and single line to ground fault of a machine terminal is used in a 3-phase inverter. In another study [40], the OC faults of thyristors used in a 3-phase full-bridge rectifier in 21 types have been investigated via the support vector data description and SVM techniques. A multi-layer ANN based on multi-valued neuron with a complex QR-decomposition has been designed and utilized in [41] to identify capacitor faults in a Class-E DC-AC inverter. In [42], four type of converter faults (namely backfire, fire-through, commutation failure, and misfire) used in an HVDC transmission system are identified by a wavelet-based ANN. A hybrid model of Park's vector

transform, DWT, and ANN methods have been presented in [43] to identify single and multiple OC switch faults under variable load conditions in a 3-phase voltage source inverters. In another valuable study [44], the OC and SC faults diagnosis in a 3-phase inverter circuit has been performed using optimized machine learning algorithms. In this paper, wavelet and PCA graph methods are selected for statistical processing of measured signals, and fuzzy logic system and relevance vector machine methods are utilized to fault detection and classification. So that, diagnostic methods are optimized using evolutionary particle swarm optimization and cuckoo search optimization algorithms. In [45], an unsupervised data mining technique called weighted Kernel PCA has been introduced and utilized to detect the OC switches faults in the 3-level inverter.

In 2016, another powerful machine learning technique called the decision tree has been proposed for the first time in [46] to detect the OC fault tested in a voltage source inverter of induction motor drives. In [47], an improved ANN algorithm called kernel SOM has been proposed to detect the SC fault in 3-phase converter-fed induction motors. Identification of AC filter's health status based on the opening/closing current of AC filter's breaker in an AC filter in converter station in [48] has been detected via the RBF neural network. In [49], an active semi-supervised fuzzy clustering algorithm with pairwise constraints has been utilized to detect the fault of the OC switches faults in a multiphase multilevel neutral point clamped (NPC) converters in a five-phase machine. In [50], the identification of parametric faults (the parametric degradation trends of resistors and capacitors from accelerated life tests) in a benchmark Sallen-Key filter circuit and a DC-DC converter system is performed via a kernel learning-based procedure. In another valuable study [51], unsupervised techniques called PCA and Kullback-Leibler divergence have been employed to detect incipient bias fault and incipient ramp fault in an inverter used in China Railway High-speed 2.

Over the past years, the use of ANN and machine learning techniques in power electronic applications has shown significant growth potential. However, advances in science and the complexity of power systems and the increasing volume of monitored data from industrial equipment and power systems have increased the need for feature extraction and pattern recognition methods. In 2017, for the first time, deep learning techniques were used to fault detection in PESs [7]. In this paper, the deep belief network as one of the deep learning algorithms has been suggested to identify hard faults such as OC and SC faults but also the soft faults such as the component degradation of power MOSFET, inductor, diode, and capacitor in a DC-DC converter (closed-loop single-ended primary inductance converter). The method proposed in this paper was optimized using the crow search algorithm. In another valuable work in 2017 [52], one of the other deep learning applications called, Sparse Autoencoder has been introduced to detect the OC fault in various modes of a cascaded H-bridge seven-level converter. Hard and soft faults in the super-buck converter circuit have been investigated in [53] using the Kernel Entropy-Based Classification approach and ELM methods. In [54], one of the novel deep learning applications called Long Short-Term

Memory (LSTM), a prominence version of the Recurrent Neural Network (RNN) has been suggested for fault detection and scalable reliability in high-frequency Gallium Nitride power dc-dc converter. In 2018, in a valuable study [55], one of the powerful applications of deep learning in feature extraction called Convolutional Neural Network (CNN) has been proposed and utilized for the first time to detect OC fault in modular multilevel converter (MMC). In [56], eight kinds of DC bus capacitor faults and energy storage inductor in dualbuck bidirectional DC-AC converter have been investigated by fuzzy cerebellar model neural network. Six different fault situations in a PWM DC-DC converters using multilayer multivalued neuron neural network have been identified in [57]. In another valuable study [58], various machine learning techniques called k-nearest neighbors (k-NN), Bagging, AdaBoost, MLP, SVM, and Naive Bayes have been employed to identify half broken rotor bar and broken rotor bar in an induction motor. In that study, the performance of each of the methods used is evaluated and compared, and finally Naive Bayes and Bagging methods are selected as the best models. The Deep CNN technique in [59] investigates the SC and OC faults in MMCs. In another study [60], a hybrid machine learning technique called mixed kernel support tensor machine detects the OC fault in a MMC. The IGBT OC fault detection in a traction inverter has been performed using a combination of WT and SVM methods in [61]. In [62], a hybrid solution based on wavelet packet and ELM techniques, which is also optimized using the Firefly algorithm, has been proposed to detect the OC switch fault in a phase shifted full bridge converter. The Sparse Autoencoder based Deep Neural Network method has been selected as one of the hybrid applications of deep learning in [63] to detect the OC fault in a 3-phase full-bridge rectifier. The OC fault associated with a Permanent Magnet Synchronous Generator Wind Energy Converters and Modular Multi-level Converter has been investigated in [64] and [65], respectively, using neural network-based techniques. The OC faults and current sensor faults in grid-tied 3-phase inverters have been identified in [66] by presenting a method that innovatively combines two types of diagnosis variables, line voltage deviations and phase voltage deviations. A hybrid method based on integrating the wavelet packet transform and LSTM has been introduced in [67] to Identify SC and OC faults in a five-level nested neutral-pointpilot topology. In [68], the CNN technique has been used as one of the deep learning applications to identify OC faults in the back-to-back converter in permanent magnet synchronous generator-based wind generation system. In another valuable study [69], the CNN method has been selected as a diagnostic tool to detect inverter faults in the PV system and symmetrical/unsymmetrical faults in the distribution line. In [70], the CNN procedure has been represented as a powerful tool to diagnosis OC switch fault in a hybrid active NPC inverter. The ELM and Random Vector Functional Link network techniques, as machine learning applications, identify and classify OC fault in an IGBT utilized in a 3-phase PWM converter [71]. In a novel study in 2020 [72], detection of multisensor-based traction converter faults has been performed

by the LSTM technique in an experimental setup. The proposed LSTM in this study, extracts the long-term patterns and dependencies in time-series effectively, and learns hidden fault features from traction converter multisensor signals adaptively, without needs of expert knowledge or system modeling. In [73], an unsupervised learning approach based on PCA has been suggested for detecting semiconductors and modules faults in Silicon Carbide MOSFETs. Moreover, in order to increase the accuracy and reduce the time horizon of abnormally detection, a PCA-based pre-processing approach is applied to the measured signals from the system.

As reviewed in the above literature, today deep learning applications based on their high capabilities in pattern recognition and feature extraction are mainly utilized in fault detection applications of PESs. Nowadays, researchers are dramatically improving deep learning methods or looking for new ones. In 2021, for the first time in [74], another deep learning technique called Temporal Convolutional Network has been introduced and employed to identify and classify OC faults and six unknown faults in a 3-phase voltage inverter.

A review of fault detection studies in PESs showed that learning-based techniques were mainly used for this purpose. In some other studies, using techniques based on hardware equipment, control, and mathematical calculations, faults in PESs have been identified [75]–[82]. Ref. [75] introduces a compensation control method based on a mixed switching strategy to detect the OC switch faults in a boost DC-DC converter. In another valuable study [76], the fault fast switch fault in a AC-DC Converters of Hybrid Grid Systems has been diagnosed using the threshold point calculation method.

A review of various studies shows that the faults related to the PESs occur mainly in industrial equipment, which increases the need for rapid detection to prevent serious damage to the system and power electronic instruments. Based on literature reviewed, the ANN algorithms have been the most widely used to detect PES faults. Since 2009, machine learning methods have come into play with the improvement of problems related to ANN techniques and their use continues. Deep learning techniques based on their high ability to extract features and high performance speed have been utilized to detect PES faults since 2017 and the use of these methods is expected to increase in the coming years.

A review of the literature in this section shows that the studies performed to detect the PESs faults have mainly focused on the detection of hard faults. It is important to note that hard faults are caused by soft faults, and if diagnostic methods focus on timely detection of soft faults, the hard faults can be prevented.

III. FAULT DIAGNOSIS METHODS IN PESS

The PESs are critical components in the power/energy systems and industrial equipment that ensure the stability and efficiency of these systems. Therefore, the health of PESs must be fully guaranteed and any abnormalities in these systems must be detected and corrected in a timely manner. Choosing the suitable method for fault detection in PESs is very important. The method selected must have the ability to act very quickly and with high detection accuracy. As reviewed in the literature, many methods for fault detection in PESs have been introduced and used so far. The continuation of this section categorizes and introduces all of the model-based and data-based fault detection techniques in PESs and finally evaluates the performance of the methods used, technically. The block diagram in Fig. 2 categorizes the types of learning-based algorithms of data mining techniques.

A. Model-based techniques

The model-based fault detection approaches have long been widely used in PESs. The performance of these techniques is based on physical processes and interactions between system components. Thus, fault detection in the system is based on the impact of physical changes in the converter model or any PES's component. The use of model-based techniques requires knowledge of the structure of a system and the characteristics of its components [7], [83]. Model-based techniques are divided into two classes: Qualitative and quantitative procedures. The qualitative-based methods do not require accurate numerical models and this is the reason why they are more resistant to noise and modeling errors. However, these model-based methods are not very accurate and do not have the ability to accurately determine the magnitude of the fault. However, quantitative methods include fault detection and the ability to determine the magnitude of the fault. Therefore, quantitative methods can be utilized for fault prognosis applications as well as estimating the useful lifetime of the PESs [84]. So far, fault detection in PESs has been performed in many studies based on model-based techniques.

In [85], quantitative fault detection has been performed by proposing a diagnostic model-based technique called a hybrid bond graph (HBG). A hybrid circuit model, based on the HBG and global analytical residuals redundancies has been introduced for fault detection in [86]. A model-based fault detection technique is proposed in [84] to detect SC faults in switches as well as incipient and abrupt faults in switches and detectors on a dc-ac half-bridge inverter. In this study, the proposed model is modeled based on the HBG and the residues. In [87], a model-based technique for detecting OC faults in a single-phase DC/AC converter, which includes an H-bridge and a capacitor with parallel resistance and current source on its DC side, has been presented. The technique proposed in this study is based on dynamic regressor extension and mixing, and works



Fig. 2. Classification of types of learning-based algorithms of data mining techniques

based on fault signature estimates. The OC fault detection in a two-level three-phase converter has been done in [88] by providing a hybrid model-based and data-based approach. The performance of the model proposed in this study is based on parameters and observations related to output currents, grid voltages, and DC voltage. A model-based state estimator procedure has been suggested in [89] for OC fault diagnosis in switches of a nanogrid prototype with a 380 V DC distribution bus. This nanogrid consists of four various switching power converters, including a buck converter, an interleaved boost converter, a single-phase rectifier, and a three-phase inverter. In [90], the OC fault detection of single-phase three-level neutralpoint clamped converters used in an electric railway has been performed by introducing a model-based fault detection approach. The proposed algorithm detects the faults using the signals existing in the control system. In order to identify the single-phase PWM rectifier open switch faults, a model-based approach based on the mixed logical dynamic model and residual production has been proposed in [91].

A review of the literature shows that model-based fault detection techniques have been widely used in the industrial applications of PESs. However, these techniques suffer from high dependence on the model and physical behavior of the system. Thus, issues such as the types of harmonics and component exhaustion can cause misleading changes in the fault pattern and complicate the fault detection process. In addition, in many cases, it is very difficult to calculate accurate mathematical representations, and the mathematical modeling of the physical model of the system under study is very complex [7]. Therefore, nowadays, in valuable studies [15], [92], the use of data-based techniques that have no dependence on the physical model of the system and do not have any computational complexity has been suggested.

B. Artificial Neural Network (ANN)

The ANN has been one of the most prominent areas of research for the past few decades and is growing rapidly nowadays. ANN is also referred to as artificial intelligence which is a system derived from human intelligence and based on training that is utilized to analyze and process various types of data [93], [94]. So far, various algorithms have been introduced for ANN that are used for applications such as regression, classification, and pattern recognition in various scientific and industrial fields. Applications such as maximum power point tracking (MPPT) control for wind power conversion systems [95], optimal and intelligent control in power electronic converters [96]-[99], intelligent controller for light emitting diode (LED) [100], [101], remaining useful life estimation for super-capacitors [102], and the identification of a variety of anomalies are considered to be the capabilities of ANN algorithms in PESs. A valuable review paper [15], fully introduces and reviews the applications of neural networks in PESs. Due to the fact that this paper mainly emphasizes the introduction of techniques used in the identification of PES anomalies, the ANN techniques used in this regard are classified and introduced as follows:

1) Multilayer Perceptron (MLP)

The MLP is one of the ANN algorithms with a layer-by-layer feed-forward structure that can mainly model different functions to solve many complex problems. In addition, solving problems such as regression, categorization, and non-linear modeling are other applications of MLP [103]. As shown in Fig. 3, the input layer (x), hidden layer, and output layer (Y) forms the structure of this network, respectively. The first layer receives the inputs and transfers them to the next layer for processing. Operational calculations and determination of weight (W) and bias (b) for data are performed in the hidden layer. The MLP uses a supervised learning procedure named back propagation for training. After completing the calculations in the hidden layer, the output layer can finally provide the desired estimation for n input samples as follow [104], [105]:

$$Y = f(b + \sum_{i=1}^{n} w_i x_i)$$
 (1)

6

One of the most important issues in achieving the ideal prediction by MLP is determining the number of neurons in the hidden layer. The MLP network training is done in a supervised-based manner with a network called backpropagation [106].

2) Self-Organization Map (SOM)

SOM is one of the ANN algorithms with supervised and unsupervised learning capability, which was first proposed in 1982 by Kohunen [107]. The SOM analyzes and processes data based on the mapping of high-dimensional data in a lowdimensional network while preserving the inherent nature of the data. This algorithm is mainly used in prediction, classification, clustering, and data visualization applications [108], [109].

The process of SOM performance, like other ANN-based techniques, is based on two modes of training and mapping. In the first stage of the SOM implementation process, the input dataset is transformed into a low-dimensional dataset (map space) during the training process. Then, low-dimensional data is classified based on the Euclidean distance mapping.

The map space consists of components called neurons or nodes arranged in a two-dimensional rectangle or a hexagonal



Fig. 3. Block diagram of MLP

grid. The number of nodes and their arrangement are determined in advance and according to the larger objectives of data analysis and exploration. Each node in the map space is associated with a weight vector. The data related to the map space is classified based on the weights of each node and their Euclidean distance on the feature space.

Mapping complex high-dimensional relationships between input and output into a low-dimensional space while maintaining the topological structure of the original data is one of the most important features of SOM. Studies have shown that SOM is one of the best learning algorithms in text clustering research. The simplicity of SOM is one of its salient features compared to other two- or multi-layer neural networks. As Fig. 4 shows, in a simple SOM topology each input neuron is directly connected to the output neuron [110].

3) Adaptive Neuro Fuzzy Inference System (ANFIS)

The ANFIS was first introduced in 1993 by Jang as a combination of ANNs and fuzzy inference to solve complex problems and to estimate nonlinear relationships between input and output functions [111]. In this hybrid model, the fuzzy part establishes the relationships between the input and output variables. Meanwhile, fuzzy membership functions become more efficient with the help of neural networks. An ANFIS model uses the Takagi-Sugeno fuzzy inference system to form a feed forward network of five layers. In this structure, the desired output is calculated based on parameters that are adjusted by the learning algorithm to minimize modeling error [112], [113]. In such network, estimation of the parameters of the membership functions is also done by the backpropagation or a mixture of backpropagation and least-square. The ANFIS is a supervised training network used primarily for modeling nonlinear functions, classification, regression, and estimating chaotic time series [114]. The training process of ANFIS network is completed in a two-step approach. The default parameters are trained via the gradient descent and, in the backward pass, by the back-propagation algorithm.

In the ANFIS structure, the input layer, called layer 0, consists of n nodes, where n is the number of inputs. The next layer is layer 1 and is called fuzzification layer, in which each node denotes a membership value as a Gaussian function with average as presented in (2):

$$\mu Ai(x) = \frac{1}{1 + \left[\frac{x - ci}{ai}\right]^{2bi}}$$
(2)

where *ai*, *bi*, *and ci* represent the parameters of the function and their values are matched in the learning phase by a back-propagation algorithm. At each step, as the parameters change, the membership function of the linguistic term *Ai* changes.

In layer 2 of the ANFIS structure, the multiplication operation for each node is represented by the strength of the rule. Thus, to find the firing strength of a rule in which the variables x_0 have a linguistic value of Ai and xi has a linguistic value of Bi, the membership values denoted by $\mu Ai(x_0)$ and $\mu Bi(x_1)$ are multiplied in the antecedent part of Rule *i*. The number of rules



in layer 2 is represented by pn nodes. Thus, n and p represent the number of input variables and the number of membership functions.

$$wi = \mu Ai(x_0) * \mu Bi(x_1) \tag{3}$$

Layer 3 is called the normalization layer and normalizes the strength of all rules based on the following equation:

$$\overline{W}i = \frac{Wi}{\sum_{j=1}^{R} Wj} \tag{4}$$

where wi demonstrates the firing strength of the *i*-th rule. This layer contains p^n nodes.

Layer 4 in the ANFIS structure is a layer consisting of adaptive nodes. Each node in this layer calculates a linear function in which the leading multilayer feed-forward neural network error function is used to adjust the coefficients of the function as follows:

$$\overline{W}ifi = \overline{W}i(p_0x_0 + p_1x_1 + p_2)$$
(5)

pi's represent the parameters where *n* denote the number of system inputs and i = n + 1. Finally, \overline{W} is the output of layer 3. The back-propagation algorithm is used as the training algorithm in ANFIS and the parameters are updated with a learning step.

Layer 5 is the output layer that the net sum of the output of the nodes in layer 4 expresses its function, and finally, the output is expressed as follows:

$$\sum_{i} \overline{W} i f i = \frac{\sum_{i} w i f i}{\sum_{i} w i}$$
(6)

where the output of node *i* in layer 4 is expressed as $\overline{w}ifi$. Finally, the overall output of the ANFIS is based on the summation of the consequences of the rule.

In recent years, in addition to fault detection in PESs, the ANFIS has been employed in other applications related to power electronics and industrial electronics, such as control, modeling, estimation, and harmonic elimination [113], [115]–[117].

4) Radial Basis Function (RBF)

The RBF was introduced in 1988 by Broomhead and Lowe as one of the generalized structures of feed forward ANNs [118]. In this type of neural network, biological neurons have a local response. Like other ANN algorithms, the RBF has a structure with an input layer, a hidden layer, and an output layer. In the input layer, the number of nodes is equal to the number of input dimensions. In the second layer, which is the hidden layer, the number of nodes depends on the complexity of the problem. In the third layer or output layer, the number of nodes is equal to the output data dimension [119]. In this structure, the input layer distributes the normalized input variables to the hidden units of the hidden layer. An RBF associated with a center vector with dimensions equal to the number of input variables is implemented by each hidden unit. In the RBF structure, the output layer is connected linearly to the hidden layer. Thus, the simple structure of RBF has made this algorithm faster and more efficient than algorithms such as MLP [120].

The orthogonal least-squares algorithm, clustering and gradient-based algorithm can be named as the most widely used RBF network training algorithms [121]. The overall output of an RBF network for a dataset D containing N patterns of (x_p, y_p) is expressed as y_p , where x_p is the input samples. The output corresponding to the *i*-th activation function \emptyset_i in the hidden layer is computed as follows:

$$\phi_i(\|x - c_i\|) = \exp(-\frac{\|x - c_i\|^2}{2\sigma_j^2})$$
(7)

where $\|.\|$ shows the Euclidean norm, σ_j and c_i represents the width and center of the hidden neuron *j*, respectively. Finally, the output associated with the node *k* of the RBF output layer is calculated as:

$$y_k = \sum_{j=1}^n w_{jk} \phi_j(x) \tag{8}$$

The RBF is mainly used for classifying, forecasting, and estimating the relationship between input and output variables. However, the network has been used in recent years for issues such as control, stability, and anomaly detection in power electronics [122]–[124].

5) Fuzzy Neural Network (FNN)

A fuzzy system performs the control and estimating process by mapping the fuzzy sets in input product hypercube to the fuzzy sets in an output hypercube. Fuzzy systems behave like associative memories that associate output fuzzy sets with input fuzzy sets [125]. Using the concept of fuzzy system, a fuzzy neural network (FNN) can be created for forecasting, categorizing, and mapping applications between input and output variables. The FNN has the advantages of fuzzy logic and neural networks. It is able to combine fuzzy reasoning in the management of uncertain information and the ability of ANNs to learn from the process. The training process of FNN network is performed by using back-propagation and gradient algorithms.

In FNNs it is based on the fact that input represents a precondition and output is considered as the result of a rule. An FNN, in addition to all its control and training capabilities, suffers from limitations such as static problems in the scope of application due to the limited advanced network structure and poor performance in large and time-series data processing [126]. However, these systems are mainly used in power electronics applications such as anomaly detection, control [127], [128], and controller design [97].

Literature review provides an overview of the performance of ANN-based techniques in power electronics applications. It is observed that the ANN-based algorithms have been widely used in power electronics industry and science. Given the focus of this paper on fault detection in PESs, Table I categorizes fault detection studies in PESs based on the ANN algorithms.

C. Machine Learning

Recently, machine learning in particular has become a highly active research field as well as an essential technology. Machine learning techniques have been able to suitably solve problems related to various scientific and industrial applications. The continuation of this section introduces the various machine learning techniques that are mainly used in power electronics applications.

1) Support Vector Machine (SVM)

The SVM is one of the supervised machine learning techniques that was first introduced in 1995 [129]. The SVMs were introduced specifically to solve problems related to classification and then generalized as support vector regression for use in linear regression problems. In general, categories related to two or more variable classes, estimation, and pattern recognition are various applications of SVM. Additionally, the SVM can globally evaluate any multivariate function with any level of approximate accuracy [130], [131]. As Fig. 5 shows, the main idea of SVM is to estimate an optimal hyperplane as the decision surface and to maximize the edge of isolation between the two data types. Finding the suitable hyperplane and predicting each sample in the corresponding class can be accomplished by training the SVM model on the training dataset, a process that involves sequentially optimizing an error function [132].

Creating a linear mapping for the $Z = \{X_i, Y_i | i = 1, 2, 3, ..., n\}$ dataset by SVM is based on the following relation:

$$\gamma = \omega^T \theta(x) + b \tag{9}$$

where ω and b represents the weight vector and bias. $\theta(x)$ denotes the agent of a nonlinear mapping function. Computation

			STUDIES THAT H	AVE IDENTIFIED FAULTS BASED ON AP	NN TECHNIQUES	
Method	Ref	Year	Fault type	PES application	Advantages	Limitations
MLP	[22]	2003	Single commutation	3-phase cycloconverter drive	Capable of implementing	The effectiveness of
			failure, Double not	scheme	on complex nonlinear	independent variables from
			successive commutation	HVDC converter	problems and data.	dependent variables is not
			failure, Double successive	Multilevel-inverter drive	Fast performance after	known in this method.
			commutation failure		training and save in the test	Computations of this
	[29]	2009	SC	Converter-fed induction motors	stage.	method is complex and
	[31]	2011	Rotor fault	Converter-fed induction motor	Ideal and accurate	time consuming
				and changeable rotors	performance against small	Model performance and
	[148]	2012	OC switch fault	Three-parallel converters in a	datasets.	test results are highly
				wind turbine	It's very simple structure	dependent on the quality of
	[149]	2013	OC	Proton exchange membrane fuel	makes it easy to design the	the training process
				cell and DC-DC Converter	desired network.	In processing high
	[33]	2013	Intern-turn SC	Three-phase converter-fed	-	dimension data, it mainly
				induction motor		suffers from over-fitting
	[35]	2014	SC incipient fault	Three-phase squirrel-cage	-	problems
	[150]	2015	Four different levels of	induction motor fed by a		Does not have the ability to
			switches fault	sinusoidal PWM converter		model time-series data and
				Series hybrid electric vehicles		extract correlations
	[57]	2018	Boundary conduction	PWM DC-DC converters and	-	between input and output
	L 1		mode, discontinuous	their applications for the buck and		variables in this data
			conduction mode, CCM.	boost DC-DC converters		
			deviations in nominal			
			values of capacitor, power			
			inductor, and duty cycle			
	[151]	2018	Stator SC	Three-phase induction motors	-	
	[29]	2019	SC	Converter-fed induction motors	-	
	[65]	2019	Single-Submodule OC	Modular multi-level converter	-	
	[05]	2017	fault	Woddia man lever converter		
BDNN	[21]	2002	OC and SC	3 phase cycloconverter drive	Correction of trajectories in	Mainly in solving most
DI ININ	[21]	2002	oc and se	scheme	weight and bias space	problems it has a high
				Duals converters applied in DC	through gradient descent is	dependence on the type of
				drives	one of the most important	inputs
	[22]	2012	Faulta in a single thuristor	Three phase full bridge controlled	features of this technique	It is highly sensitive to
	[32]	2012	and the faults happening	roatifier	Due to the removal of	complex and poisy data
			in two thuristors at the	lectifier	weight links it provides a	Cannot do time-series data
			in two unyristors at the		very simple network	modeling
	[33]	2013	Intern turn SC	Three phase converter fed	structure	The processing of large
	[33]	2015	intern-turn SC	induction motor	It has fast and easy	volumes of data by this
	[52]	2017	OC awitab fault	Cascadad H bridge seven level	programming	method suffers from the
	[32]	2017	OC switch fault	Cascaded H-blidge seven-level	No prior knowledge of	problem of over-fitting
	[152]	2017	Diata OC		networks is required	Not able to extract features
	[152]	2017	Diode OC	Inree-phase full-bridge rectifier	Its training process is	from input data
	[/]	2018	OC, SC, component	DC-DC Converter (closed-loop	independent of the features	nom mpår data.
			degradation of power	single-ended primary inductance	of the function	
			MOSFET, inductor,	converter)	Allows efficient calculation	
	1.4.63	2010	diode, and capacitor	R 11 1 111 1 1 R 1 R 1 R 1 R 1 R 1 R 1	the gradient in each layer	
	[56]	2018	Eight kinds of faults	Dual-buck bidirectional DC-AC	completely	
			related to DC bus	converter	completely.	
			capacitor and energy			
			storage inductor	20/2		
	[59]	2018	Switch OC and SC	MMC	-	
	[61]	2018	IGBT OC	Traction inverters	<u>.</u>	
	[153]	2019	Multiple OC switch fault	A back-to-back converter in		
				doubly-fed induction generator-		
				based wind turbine systems		
SOM	[26]	2007	Driver power supply	Isolated DC-DC converters in	The process of test and	The process of training a
			under voltage, transistor	multi-phase multi-level motor	evaluating new data after	network to deal with high-
			SC, and overcurrent	drive	network training is very	dimension data is time-
	[33]	2013	Intern-turn SC	Three-phase converter-fed	fast.	consuming.
				induction motor	The SOM has a very simple	Despite processing large
	[47]	2017	SC	Three-phase converter-fed	network structure that	data, it does not have the
				induction motors	avoids the complexity of	ability to model the time-
					computing.	series mode of data.
					It can also be used as an	The training process of this
					unsupervised procedure.	network and its
					It can be used as a tool to	performance in the test
					dimension reduction of	phase is highly dependent
					high-dimension data.	on the quality of the input
					U	data.
RBF	[154]	2003	OC	Inverter drive		

TABLE I SI ANN

9

	[31]	2011	Rotor fault	Converter-fed induction motor and changeable rotors	It has a fast process of determining the network	The large number of neurons increases the
	[60]	2018	Switch OC	MMC	parameters and training	complexity of the network
	[64]	2019	OC	Permanent magnet synchronous generator wind energy converters	stage. The RBF has a high ability	for processing input variables and determining
	[64]	2020	OC in both single and double switches	A permanent magnet synchronous generator system for wind turbines	to solve function approximation problems for data and surfaces with regular peaks and valleys. Ideal performance and high resistance against noisy data are the prominent features of this technique. Fast performance after training in the test stage.	their correlation with output variables. The RBF network training algorithm is incapable of processing and modeling robust and complex nonlinear systems. Like other traditional neural networks, it suffers from time-series data modeling and high- dimension data.
FNN	[21]	2002	OC and SC	Duals Converters Applied in DC Drives	This technique has a much better and more accurate	The limited structure of this network causes limitations
	[23]	2004	Fault-free circuit	Delta-sigma converter	learning ability and the	such as static problems in its application areas. High-dimension data
	[27]	2007	OC and SC switch faults	A dc-motor-based brake-by-wire system	convergence error in this network is very small.	
	[44]	2016	Transistor OC	Three-phase inverter circuit	Compared to other ANN	processing is not very
	[155]	2017	Structural and functional faults	Analog to digital converter	techniques, it has a high ability in modeling and	accurate and it is not possible to discover the
	[49]	2017	Switch OC	Multiphase multilevel NPC converters in five-phase machine	mapping nonlinear systems.	correlation between input and output variables in
	[56]	2018	Eight kinds of faults related to DC bus capacitor and energy storage inductor	Dual-buck bidirectional DC-AC converter	The structure and training process of this network requires less adjustable support parameters than other ANNs. Better integration of this network with other control design methods is a prominent feature of this method.	ture and training of this network less adjustable parameters than Ns. tegration of this with other control methods is a t feature of this



Fig. 5. Main idea of SVM

of support vectors associated with each class are described as:

$$\begin{cases} b + W^T X_i = +1, & for \ d_i = +1 \\ b + W^T X_i = -1, & for \ d_i = -1 \end{cases}$$
(10)

where d_i shows the related class, i.e., $d_i = +1$ of class A and $d_i = -1$ related to class B.

In the SVM structure, the transfer of inseparable data to a high-dimensional linear space and their classification based on the linear hyperplane is possible using a vector mapping function $\varphi(x)$. Finally, the implementation of the decision function is done as follows:

$$f(x) = sign\left(\sum_{i=1}^{N} a_{0,i}\left(\varphi(x)\varphi(x_i)\right) + b\right)$$
(11)

The training process of the SVM network is based on different kernels. Various types of kernels such as linear, nonlinear, and polynomial can be used as SVM training algorithms.

As mentioned in the literature, the SVM has been used extensively in recent years to identify faults related to PESs. However, its use is not limited to this topic and it has been utilized in other applications such as estimating the batteries state of charge, reliability, and control issues [133]–[136].

2) Extreme Learning Machine (ELM)

The ELM was first introduced in 2006 as one of the machine learning applications and learning tool based on a modification of the traditional single hidden layer feed-forward neural network [137]. The ELM technique is mainly used for classification applications, regression-based forecasting in short-term intervals, and estimating the relationship between input and output variables. As Fig. 6 shows, the ELM structure is consisting of the input layer, hidden layer, and output layer [138]. Unlike the ANN algorithms and some machine learning techniques, the ELM has a very fast training process and



Fig. 6. Schematic diagram of a ELM network

randomly selects hidden thresholds during the training period and analyzes the output weight without iterative calculations.

The ELM as an efficient tool has unique advantages such as higher performance and suitability for kernel functions and nonlinear activation. The best feature of the ELM algorithm is its very high training speed, which is based on the simplicity of its structure [139], [140].

The gradient descent-based back-propagation training algorithm is most widely employed for the training of ELM [141]. In the first step of ELM, the network parameters are assigned randomly. Then, the output matrix related to the hidden layer calculation for the input weight vectors $x_i = [x_{i1}, x_{i2}, ..., x_{iN}]^T$ is calculated as follows:

$$\sum_{i=1}^{\tilde{N}} \beta_i f_i(x_i) = \sum_{i=1}^{\tilde{N}} \beta_i f(a_i \cdot x_j + b_j) = t_j, \qquad j = 1, \dots, N$$
(12)

where b_i denote the hidden layer bias and *i* is the number of hidden layer neurons, and the hidden neurons are assigned as \tilde{N} . The weight vectors which connect the input nodes and *i*-th hidden layer nodes are shows as $a_i = [a_{i1}, a_{i2}, ..., a_{in}]^T$. The output weight vectors that connect output layer neurons with the *i*-th hidden neuron is demonstrated as $\beta_i = [\beta_{i1}, \beta_{i2}, ..., \beta_{in}]^T$. In the ELM architecture $f(a_i, x_j + b_j)$ is the activation function and is represented as follow:

$$f(a_i \cdot x_j + b_j) = \frac{1}{1 + e^{-(a_i \cdot x_j + b_j)}}, \quad i = 1, \dots, L, \quad (13)$$
$$j = 1, \dots, N$$

The equation (12) can be written as follows [142]:

$$H\beta = T \tag{14}$$

with

$$H = \begin{bmatrix} f(a_1 \cdot X_1 + b_1) & \cdots & f(a_{\bar{N}} \cdot X_1 + b_{\bar{N}}) \\ \vdots & \ddots & \vdots \\ f(a_1 \cdot X_N + b_1) & \cdots & f(a_{\bar{N}} \cdot X_N + b_{\bar{N}}) \end{bmatrix}_{N \times N}$$
$$\beta = \begin{bmatrix} \beta_1^T \\ \vdots \\ \beta_N^T \end{bmatrix}_{N \times m} \quad T = \begin{bmatrix} t_1^T \\ \vdots \\ T_N^T \end{bmatrix}_{N \times m}$$

where *H* is the hidden layer output matrix, β and *T* show the output weight matrix and the matrix of target output, respectively. Finally, the input weight and bias in the training stage are generated randomly, and the output weight can be calculated by using the Moore-Penrose generalized inverse of *H* as follow [142], [143]:

$$\beta = H^{\dagger}T \tag{15}$$

This technique has been used since 2013 to detect faults in PESs and has since been used in the power electronics industry for applications such as design, control, condition monitoring, and nonlinearity mitigation for light-emitting diode communications [144]–[147].

3) Decision Trees (DT)

The decision tree is a type of inductive learning and is known through a systematic method known as recursive binary partition. This technique can be used in different application ranging from linear regression models, classification, and feature extraction, to pattern recognition [156], [157]. As Fig. 7 shows, the structure of decision trees consists of tree-shaped diagrams [158]. Thus, this tree is formed from components called branches and three types of nodes called root node, internal node and leaf node. In this structure, the dataset is divided into a number of subsets through a series of dichotomous classifications. In the interconnected structure of the decision tree, nodes are the points in the tree where features are processed. Branches are also test results that form the next nodes. Among the nodes, the root node is the highest node, the internal nodes in the middle and the leaf nodes at the end are known as the end nodes. The decision tree training and testing process ends when a node reaches a certain predetermined class purity level and there is only one output type in that node [158], [159]. In some reviewed studies, the decision tree technique has been used to identify PESs anomalies. In some other valuable studies [151], [157], [160], [161], these techniques have been used in various applications of power electronics and industrial electronics. After introducing each of the machine learning techniques and introducing their various applications in power electronics, the classification of fault detection studies in PESs based on the machine learning algorithms is performed in Table II.

D. Deep Learning

The complexity and nonlinear behavior of PESs have increased the need for accurate and high-performance methods in fault detection operations. Although the ANN and machine learning



Fig. 7. Schematic diagram of a decision tree

			STUDIES THAT HAVE IDEP	TIFIED FAULTS BASED ON MACHINE I	ARNING TECHNIQUES		
Method	Ref	Year	Fault type	PES application	Advantages	Limitations	
SVM	[30]	2009	OC switch fault	PWM voltage fed power converter of brushless DC motor drive	The high impact of this method on high- dimensional data is a	The SVM algorithm is not suitable for large data sets. In the face of noise data	
	[32]	2012	Faults in a single thyristor and the faults happening in two thyristors at the same time	Three-phase full-bridge controlled rectifier	prominent feature. In data where the number of features is more than the number of rows of data,	that is closely correlated between target classes, it does not perform as well. In cases where the number	
	[34]	2013	valve SC, valve pulse loss, single-phase grounding, two-phase grounding, three-phase grounding, DC line grounding	HVDC converter	SVM can provide an acceptable application. High ability in classification applications even for multi-class data	of features of each data point exceeds the number of training data samples, SVM will perform poorer. Choosing an optimal kernel	
	[35]	2014	SC incipient fault	Three-phase squirrel-cage induction motor fed by a sinusoidal PWM converter	based on creating a hyper- plain that separates the features of each class well.	for SVM is one of the main problems of using this method.	
	[37]	2015	Switch OC	Cascaded H-bridge multi-level (5-level) inverter	regression and estimation	the face of time-series data	
	[40]	2015	thyristors OC fault	Three-phase full-bridge rectifier	applications.	is considered as one of the	
	[162]	2016	Switch OC	Three-level NPC inverter	techniques it uses much	problems of this method.	
	[152]	2017	Diode OC	Three-phase full-bridge rectifier	less memory to process		
	[155]	2017	Structural and functional faults	Analog to digital converter	 less memory to process data. 		
	[7]	2018	OC, SC, component degradation of power MOSFET, inductor, diode, and capacitor	DC-DC Converter (closed-loop single-ended primary inductance converter)			
	[56]	2018	Eight kinds of faults related to DC bus capacitor and energy storage inductor	Dual-buck bidirectional DC-AC converter			
	[59]	2018	Switch OC and SC	MMC			
	[60]	2018	Switch OC	MMC			
	[61]	2018	IGBT OC	Traction inverters			
	[153]	2019	Multiple OC switch fault	A back-to-back converter in doubly-fed induction generator- based wind turbine systems			
ELM	[33]	2013	Intern-turn SC	Three-phase converter-fed induction motor	The ELM has a very simple structure that minimizes	It does not provide acceptable performance in	
	[35]	2014	SC incipient fault	Three-phase squirrel-cage induction motor fed by a sinusoidal PWM converter	computational complexity and speeds up the training process of this network.	processing data whose number of features is more than the number of rows.	
	[163]	2016	Closed-loop	Single-ended primary inductance converter	The selection of hidden thresholds is done	Processing large data with this technique does not	
	[53]	2018	Soft and hard faults	Super-buck converter circuit	randomly so that the output	have good results.	
	[62]	2018	Switch OC	Phase shifted full-bridge converter	iterative calculations.	relationship of input data	
	[71]	2019	Switch OC	Insulated gate bipolar transistor (IGBT) used in three-phase PWM converter	and stability in kernel functions and nonlinear activations. It performs very well in the processing of nonlinear systems.	The training process in the data that the output labels are close to each other has more problems.	
DT	[31]	2011	Rotor fault	Converter-fed induction motor and changeable rotors	Compared to other techniques, decision trees	The smallest changes in the data process cause large	
	[46]	2017	OC	Voltage source inverter for induction motor drives	require less effort for data preparation during pre-	changes in the structure of the decision tree and cause	
	[151]	2018	Stator SC	Three-phase induction motors	processing. In the face of noise data, there is no need for pre- processing. They provide acceptable performance in the training process without the need for data scaling. After completing the training process, the test stage is performed at a very high speed by the saved network.	network instability. In dealing with large and high-dimension data, they often involve a lot of computational complexity. The decision tree often takes more time to model training. They do not have acceptable generalizability for regression applications and estimation of insufficient continuous	

 TABLE II

 Studies that have identified faults based on Machine learning techniques

techniques have been widely used in PESs, increasing the value of monitored data from new smart systems has also reduced the accuracy and performance of these techniques. Today, deep learning techniques are employed as algorithms with high ability to extract features and accurately analyze and model the nonlinear behavior of PESs. Deep learning has various algorithms such as the CNN, RNN, LSTM, and Auto-encoders. In the continuation of this sub-section a complete introduction of such method together with its capabilities in fault detection and other applications of industrial and power electronics are outlined.

1) Convolutional neural network (CNN)

F F

The CNN is one of the most prominent deep learning techniques, known as a powerful tool in feature extraction. This technique has been used in many scientific and industrial fields so far. Classification, feature extraction, and high volume data processing have been the most important applications of this technique [164], [165]. The CNN, by providing a layer-to-layer structure and in-deep training, has been able to cover many of the shortcomings of ANN and machine learning algorithms. Fig. 8 shows the structure of CNN which includes the input layer, convolution layers, pooling layers, fully-connected layers, and finally the classification layer (Soft Max) [166], [167]. Each layer in this structure offers a unique function until the ideal output is predicted. The input layer receives the input data and transmits it to the first convolution layer. Each convolution layer consists of kernels that act as filters on the data and extract their features. Convolution operations associated with each layer are done as follows [166]:

$$y_{ij} = \sigma(\sum_{r=1}^{N} \sum_{c=1}^{N} W_{rc} X_{(r+i \times S)(c+j \times S)} + b)$$

$$0 \le i \le \frac{Hd - Fd}{S}, \qquad 0 \le j \le \frac{W - Fd}{S}$$
(16)

where y_{ij} represent the output of each node in the convolutional layer. *Hd* and *Fd* represent the length and height dimensions corresponding to the input variables, respectively. *S* stands for stride length. W_{rc} and *b* denotes the weight and bias associated with each node. The activation function in this layer is shown as term σ . The activation function in each convolution layer of CNN is performed by the rectified linear unit (ReLU), as a nonlinear activator function, as follows [168]:



Fig. 8. Layer-to-layer structure diagram of CNN

The extracted features by each filter are collected by the pooling layer in each convolution layer and transferred to the next convolution layer as a feature map. The pooling operation in each layer can be done on two bases, average pooling and Max pooling. The pooling operation is done in each convolution layer as follows [168]:

$$x_i^l = f\left(\beta_i^l pooling\left(x_i^{l-1}\right) + b_i^l\right) \tag{18}$$

where x_j^l is the output of the *j*-th filter in convolution layer *l*; *pooling* () denote the pooling operation and β represent the pooling kernel.

Studies have shown that Max pooling can be more efficient because it contains the highest value of extracted features. This run continues to the last convolution layer until the feature map extracted from the last convolution layer is considered as the input of fully-connected layers. The fully-connected layers are a type of feed forward-based neural networks that is responsible for determining the weight and bias of the nodes in the extracted features. The CNN training process is performed in fully connected layers using an MLP neural network. In the final stage of CNN, a SoftMax layer predicts and classifies the extracted features based on designated labels [169], [170].

In the evaluation of CNN applications, various studies show that this technique has been used not only in fault detection of PESs, but also in estimating electric motor temperatures and SOC estimation of batteries [171], [172].

2) Recurrent neural network (RNN)

The RNNs are a class of deep learning techniques based on artificial neural networks used to process sequential data. Estimating the correlation between input and output variables, processing large volume data, forecasting, and classification are the various applications of RNNs [173], [174]. As shown in Fig. 9, the connections between nodes in an RNN form a guided graph along a time sequence. In addition, the RNNs in the training phase use recurrent units to store past information to use the stored information as input to subsequent layers. This recurrent process enables RNNs to perform excellently in identifying the behavioral pattern of the input data [175]. However, as the training process increases in length and the process becomes longer, training RNNs to obtain long-term data dependence becomes more difficult and requires more memory. As a result, the network encounters problems such as gradient vanishing or rarely, gradient explosion, which makes gradient-based learning methods impractical [175], [176]. To solve this problem and improve the training process for timeseries data and high-dimensions, two algorithms, LSTM [177] and gated recurrent unit (GRU) [178] have been introduced.

In recent years, the RNN techniques have been widely used as a powerful tool for a variety of applications such as converter control, harmonic prediction at nonlinear loads, maximum power factor searching, and impedance measurement in PESs and industrial electronics [179]–[181].

The LSTM is one of the most prominent versions of RNNs, which was first proposed in 1997 to improve traditional RNN performance and learn long-term features [177]. The main idea that LSTM pursues is the introduction of an adaptive gateway mechanism that determines the extent to which the memory unit retains the previous state while at the same time, it remembers the features of the current input data. This function is very suitable for processing and predicting events with long intervals and delays in time-series. Furthermore, the classification, forecasting, signal processing, and pattern recognition for highdimensional data are prominent applications of LSTM [182], [183]. As Fig. 10 shows, each LSTM unit is formed as an input gate, output gate, forget gate, and memory cell as the hidden layer. In addition, the LSTM network replaces neurons in the hidden RNN layer with a memory unit to realize the memory of past information [184].

Gradient descent and back-propagation through time are among the well-known algorithms used to training various types of RNN networks such as LSTM and Bidirectional LSTM. The back-propagation through time is a generalization of back-propagation for feed-forward networks. The mathematical formulation related to the LSTM architecture is as follows:

$$f_t = \sigma(W_{lf}l_t + W_{mf}m_{t-1} + b_f)$$
(19)

$$i_t = \sigma(W_{li}l_t + W_{mi}m_{t-1} + b_i)$$
 (20)

$$o_t = \sigma(W_{lo}l_t + W_{mo}m_{t-1} + b_o)$$
(21)

$$a_t = tanh(W_{la}l_t + W_{ma}m_{t-1} + b_a)$$
 (22)

$$c_t = c_{t-1} \phi f_t + i_t \phi a_t \tag{23}$$

$$m_t = o_t \phi tanhc_t \tag{24}$$

where σ shows the logistic sigmoid function, f_t, i_t , and o_t represent the forget, input, and output gates, respectively. c_t and a_t are the memory cell and hidden vector, respectively. $W_{l*} = \{W_{lf}, W_{li}, W_{la}, W_{lo}\}$ and $W_{m*} = \{W_{mf}, W_{mi}, W_{ma}, W_{mo}\}$ denote the trainable weights associated with respective gates. b_f, b_i, b_o , and b_a represent output biases for each gate. Finally, operator ϕ represent the Hadamard product.

Each of these gates in the LSTM structure offers a unique function to predict the final output in the best possible way. Details on the structure and function of LSTM are introduced



Fig. 9. Structural schematic diagram of RNN



Fig. 10. Block diagram of a LSTM unit

in [185]. As stated in the literature on the ability of the LSTM technique to detect PESs faults, it should be noted that this method is also used in a variety of issues such as pulsed load monitoring and tolerance control related to PESs [67], [186].

3) Autoencoders (AEs)

An autoencoder (AE) is one of the deep learning applications that is often applied to reduce the dimension of the data and extract the data features with minimal reconstruction error. As Fig. 11 shows, an AE consists of an encoder and a decoder network, which also consists of an input layer, several hidden layers, and an output layer [187], [188]. The structure and function of AE is such that based on nonlinear layers it can retain the most data-related features and the most important information after dimensionally reduction. In the AE training phase, with proper retrieval of the input data, maximum information about the original input data is stored by hidden cells in the hidden layers [164].

The encoding and decoding operations via AE for the input variable x is performed as follow:

$$f = g(\omega_x * x + b_x) \tag{25}$$

$$y = g(\omega_v * x + b_v) \tag{26}$$

where y is the output features and f denote the network features. ω_x , ω_y , b_x , and b_y represent the input-to-hidden weights, hidden-to-output weights, bias of hidden units, and bias of output units, respectively. g(.) demonstrate the activation function.

Minimizing reconstruction errors is one of the main objectives of AE training. To achieve this, the proximity between x and y must be expressed by computing the loss function (L) for *n* input samples as follows:

$$L_{AE} = \frac{1}{n} \sum_{i=1}^{n} (x_i - y_i)^2$$
(27)

Following this proceeding, the reconstruction error is computed as follows:

$$RE = \sum_{i=1}^{n} (x_i - y_i)^2$$
(28)

AE is trained based on clean x data. The test of the AE is performed using x-noise data obtained by adding noise to clean x-data. Upon completion of this process, the reconstructed output becomes as \hat{y} and L_DAE is defined as follows:

$$L_{DAE} = \frac{1}{n} \sum_{i=1}^{n} (x_i - \hat{y}_i)^2 + \frac{\lambda}{2} (\|W_1^2\|_F^2 + \|W_2^2\|_F^2)$$
(29)
+ $\|W_2^2\|_F^2$)

AEs are divided into different categories, each of which has been used in various applications in industrial electronics and power electronics [186], [189]. A review of the literature showed that in recent years, deep learning techniques have been very successful in various applications of the power electronics industry. So that these techniques were able to compensate for the shortcomings of ANN and machine learning methods and were considered a great success in identifying faults related to PESs. Table III lists the fault detection studies in PES based on deep learning techniques.

It should be noted that the significant points that should be noted are the determination of the number of epochs and the determination of the amount of data to perform the training and test steps related to supervised techniques. In learning-based data mining techniques, part of the data is considered for training and another part for test or validation the network. The determination of each of these values is the responsibility of the user and it is commonly shown in various studies that 70% to 80% of the data is selected for the training process and the rest for the test or validating the network. Determining the number of epochs or iterations related to each network can also be done in different ways. Thus, in many techniques, this process is based on the trial and error method and, in others, it is based on optimization algorithms. The most ideal number for epoch is achieved when the results of training and test processes are accompanied by the least amount of error values.



Fig. 11. The structure diagram of a Autoencoder

E. Unsupervised Techniques

PES converters have the nonlinear behavior, and in the face of some anomalies, the behavior of these systems changes significantly. Any fault or anomaly that occurs on the PES converters induces effects on the system behavior, the extraction of these effects from the characteristic curves of the system starts the process of anomaly detection [38]. As noted in the literature, intelligent learning-based techniques have been used in most studies as a variety of fault detection tools. However, in most cases, these techniques have suffered from some problems in the face of complex systems and high dimension data that contain noise with abnormal behavior in the signals received from the system.

Many studies have shown that different faults have different effects on voltage and current signals from electronic power converters, and using these signals as input to diagnostic methods requires a preprocessing operation. In signal preprocessing, the correlation between the variables is removed, the variables are reduced, and the most significant though components are extracted by weighing the various variables to somehow see the effects of the fault on the signal more clearly [73]. Unsupervised data mining applications have been used effectively to solve these problems and perform signal preprocessing to improve the anomaly detection process in PESs. Unsupervised techniques mainly include dimensional reduction techniques, which can be referred to as methods PCA [38], [73], kernel PCA [45], Kullback-Leibler [51], Isometric feature mapping [190], and feature learning [191]. In addition to fault detection in industrial electronic equipment and PESs, these techniques have been also used in other applications [165].

IV. ROUTINE OF FAULT DIAGNOSIS IN PESS

The fault detection process in PESs is based on a step-by-step procedure. In the first stage, the signals related to the healthy and damaged state of the system are measured. In the second stage, the measured signals can be normalized, filtered or preprocessed according to certain criteria in order to reduce the data dimension, and to use important parts of the signals. In the third stage, the selected diagnostic technique is designed and finally, the available signals are used as input to the desired procedure. Every PES has characteristics curves that show the performance of the system at a given time point. Each anomaly that occurs in PES has a special effect on the voltage and current signals, which most diagnostic methods used to identify faults by considering voltage and current signals as indicators. These signals are measured by sensors in the system and provided to the users. Sometimes signal measurement sensors in PESs can also be damaged while posing new problems to the system performance. Fault detection techniques must be exact and intelligent enough to detect the effects of system harmonics, sensor abnormalities, and hard and soft faults. Fault detection using voltage and current signals can also be done in the frequency domain. Thus, the signals measured in real-time are transmitted to the frequency domain using the fast Fourier transfer (FFT) to be used as input for diagnostic methods. Each

N .1 .1	D.C	17	STUDIES THAT HAVE ID		Addited Heelindgees	x • •, .•
Method	Ref	Year	Fault type	PES application	Advantages	Limitations
CNN	[55]	2018	OC	MMC	_ Use local spatial coherence	In the face of noise data,
	[59]	2018	Switch OC and SC		at the input (often images),	various pre-processing
	[68]	2019	Switch OC	Back-to-back converter in	which allows the CNN	steps must be used so that
				permanent magnet synchronous	to the sharing of some	is not reduced
				generator-based wind generation	to the sharing of some	This method does not have
	[(0]	2010	0 : 1 00	system	Benefiting from a layer by	sufficient conspility in
	[69]	2019	Switch SC	5-level neutral-point-clamped	layer structure makes them	spatially invariant to input
				to BV integrated microgrid	less computationally	data
				system	complex.	Using CNN requires a lot
	[70]	2020	Switch OC	Hybrid active NPC inverter	Excellent performance in	of training data. So that it
	[70]	2020	Power tube double	Three phase voltage inverter	feature extraction and	does not have acceptable
	[/4]	2021	open faults and power	nlatform	pattern recognition from	performance in the
			tube triple-open faults	platolin	input data and ideal for	processing of low data.
			tube utple open laans		image processing	Time-series mode
					applications.	modeling of input data is
					Contains a weight division	also not possible using this
					in layers that increases the	method.
					speed of training stage.	
RNN	[54]	2018	degradation fatigue	High-frequency Gallium Nitride	The RNN network	Suffers dramatically from
			modeling	(GaN) power dc-dc converter	remembers all the events	the problems of gradient
	[7]	2018	OC, SC, component	DC-DC Converter (closed-loop	reviewed in the training	vanishing and exploding.
			degradation of power	single-ended primary inductance	process.	RNN networks have a very
			MOSFET, inductor,	converter)	Performs data time-series	difficult training process.
			diode, and capacitor		modeling to an acceptable	No processing of long
					degree.	sequences data when using
					It can easily provide a	functions
					strong nybrid model with	The training process is
					techniques based on ANN	very time consuming and
					or deep learning	does not provide
					High-dimensional data	acceptable performance in
					processing is also a	processing noisy data
					prominent feature of these	processing noisy dutai
					networks.	
LSTM	[54]	2018	Degradation fatigue	High-frequency Gallium Nitride	Completely solves and	In time-series data when
			modeling	power dc-dc converter	eliminates problems	features are sequential,
	[67]	2019	Switch SC and OC	Five-level nested neutral-point-	related to gradient	modeling input features is
				pilot (NPP) topology	vanishing and exploding in	difficult and somewhat
	[153]	2019	Multiple OC switch	A back-to-back converter in	RNNs.	impossible.
			fault	doubly-fed induction generator-	Provide acceptable	The selection of
				based wind turbine systems	performance in large and	parameters related to the
					sequential data.	network structure is of
					modeling the time-series	great importance and the
					mode of the data and	dependent on the specified
					between the input and	parameters
					output variables in this	The training process is
					type of data	time-consuming
Autoencoder	[52]	2017	Switch OC	Cascaded H-bridge seven-level	They can be used as the	The choice of ideal model
	[0=]	_~./		converter	supervised and	parameters is tedious and
	[63]	2018	Switch OC	Three-phase full-bridge rectifier	unsupervised for a variety	has a significant impact on
	[00]			F8	of data-based applications.	network output.
					Autoencoders have an	In unsupervised
					acceptable ability to	approaches and dimension
					remove noise from data	reduction applications,
					and is ideal for real-world	some important data
					data.	information can be
					For forecasting	removed.
					applications, they can	They have a high
					easily provide a hybrid	sensitivity in the training
					model with other deep	process and the output of
					learning techniques.	the test phase is very
					Autoencoders provides	effective from the training
					acceptable performance for	phase.
					processing time-series data	Autoencoders have a high
					even when they contain	structural complexity and it
					noise.	is difficult to add or
						remove a layer.

TABLE III STUDIES THAT HAVE IDENTIFIED FAULTS BASED ON DEEP LEARNING TECHNIOUES

of the introduced methods, based on a specific strategy, detect faults based on the measured voltage and current signals. The simplest possible case is related to ANN techniques. These techniques do not have the ability to extract features and process time-series data and only detect faults based on the determination of weight and bias on the input signals. The machine learning techniques have the ability to recognize complex patterns and modeling non-linear data, but have difficulty processing signals with high volumes and dimensions. Finally, deep learning techniques with the ability of feature extraction in a layer-by-layer structure, especially for high dimension and time-series data, have been successful in timely and accurate detection of PESs faults. Fig. 12 shows the fault detection routines in PESs based on the applications of ANN, machine learning, and deep learning techniques.

V. SENSORS AND INSTRUMENTS NEEDED FOR SIGNALS MEASUREMENT

Fault detection in PESs is based on voltage and current curves. Voltage and current measurement in PESs requires instruments and sensors that are used based on their unique accuracy and bandwidth. Sensors are devices that can sense or identify and react to certain types of electrical or optical signals. There are different types of sensors related to measuring voltage and current, which are classified based on their performance and method of measurement, which the continuation of this section introduces each of them.

A. Current sensors

Current sensors are typically classified into three categories: shunts (resistance), current transformers, and Hall effect sensors. Each of these sensor classes has a different performance in terms of accuracy and bandwidth, depending on their maximum measurable frequency. However, some special sensors, such as shunts, have significant direct current measurement capabilities. While current transformers are not very capable of performing such measurements. Current measurement using shunts has several major advantages such as high accuracy, low cost (but can be more expensive for high accuracy measures), robustness, and high bandwidth.

However, this technique suffers from the lack of galvanic insulation between the power circuit and the control or monitoring circuit where the extracted signal is processed.

B. Voltage sensors

A voltage sensor is used to monitor and calculate the supply of voltage to a device. Voltage sensors have the ability to detect AC voltage or DC voltage level. The inputs of these sensors is voltage and can provide state of switches, analog current level, analog voltage signal, a current signal or an audible signal, and frequency as output. The output provided varies by type of sensor. Thus, in some of them the output can be sine or pulse trains and in others it can be presented as amplitude modulation, pulse width modulation, and frequency modulation output. Measurements in voltage sensors are based on voltage divider. Voltage sensors are classified into two types include capacitive type voltage sensors and resistive type voltage sensors.

Small size and light weight, high safety for personnel's, very high degree of accuracy, non-saturable, very wide dynamic range, ability to combining the voltage and current measurements into a single physical device with small dimensions, and eco-friendly are the important advantages of voltage sensors over the conventional measuring methods. In addition to fault detection, voltage sensors have some other important applications such as power failure detection, temperature control, load sensing, power demand control, and safety switching.

C. Feature extraction from measured signals

Measured signals from the system are utilized as input to the fault detection method because they contain the system performance and the effects of the fault on the system circuit. Techniques used in a variety of ways analyze input signals to achieve abnormal effects of faults. However, the range of the measured signal must be within the standard range. Considering low sampling rate or sampling in low range for signal measurement loses significant information about system behavior, while considering more range and high sampling rate obvious information to loss between also causes incomprehensible information. After measuring the signal, choosing the appropriate method for analysis is considered the most important step in fault detection in PESs. Each method identifies and estimates the effects of different types of faults on the measured signal based on its ability. Among these, feature extraction from measured signals can be the best approach to analyze the types of fault effects and achieve the diagnostic goal. Simply put, in the feature extraction approach,



Fig. 12. Fault detection routines in PESs based on the applications of ANN, machine learning, and deep learning techniques

unnecessary values and parameters are excluded from the sampled signal, and the effects and pattern of anomalies occurring in the system are made available as obvious and valuable information for evaluation. This approach can dramatically speed up the fault detection process in PESs and increase detection accuracy. In general, the power electronic equipment used in the system carries the noise or harmonics in the measured signal, which most fault detection methods fail to distinguish these harmonics from the effects of faults. Using intelligent techniques that have the ability to extract features can easily solve this problem and can distinguish between harmonic behavior and fault effects. Accordingly, the use of feature extraction and pattern recognition techniques is significantly recommended for fault detection applications in PESs that always have natural system harmonics and noises. Among the introduced methods, ANN algorithms determine the relationship between input variables (signals measured from the system) and output (type, location, or intensity of faults) by determining the weight and bias in the hidden layers. These techniques require a long-time and high processor memory in the face of high dimension signals and suffer from other problems such as overfitting. Machine learning techniques have improved ANN problems in terms of speed and memory, but these techniques also require proper datasets and in most cases involve overfitting problems. In the meantime, deep learning techniques have been able to solve problems related to previous solutions with a layer-by-layer structure that provides a deep learning of input data. These techniques significantly benefit from the feature extraction block in their structure and based on the extraction of prominent features from the measured signals, they detect faults. Fig. 13 shows the feature extraction procedures of signals measured. It can be seen that the fault behavior pattern in voltage or current signals is extracted and used by removing large volumes of the signal.

The penetration of deep learning techniques in power electronics applications has been able to solve major problems in this industry and on the other hand these techniques can be employed as a powerful tool in fault prognosis and online monitoring of PES. the system can continue to operate and work in the event of failure of some of its components. Stability and uncertainty in PESs is one of the most important issues that has posed many challenges in recent years. Accordingly, fault-tolerant is one of the most important factors in PESs that must be addressed. A fault-resistant design enables the system to continue to operate at the desired level, possibly at a reduced level, if part of the system fails[192], [193]. In organizing a fault-tolerant system, the first step is to diagnose and, more importantly, to prognose the fault [194]. Timely fault detection and accurate protection can prevent the fault from progressing and its harmful consequences. The next step is the fault-tolerant function, which consists of fault separation and reconfiguration. The second step is always based on the design of the hardware plugin and the control of the relevant fault-tolerant [192]. To date, many solutions to fault-tolerant operations in PESs have been proposed and utilized in various studies. The presentation of fault tolerance methods relies mainly on the type of hardware redundancy and according to this, classified into three main categories as follows: 1) switch-level, 2) leg-level, 3) modulelevel. Table IV classifies the fault tolerance methods in PESs based on these three categories.

The use of fault tolerance methods in PESs makes the system able to continue to operate for a limited time after the fault occurs. However, the performance of the system and its efficiency decreases compared to the normal state, but the effect of the fault that occurred in the system is not clearly seen. Accordingly, fault tolerance methods make many conventional techniques in the early stages unable to detect faults and the system continues to behave in a damaged state. It should be noted that under these conditions, deep learning algorithms that have the ability to extract features from the system and are able to detect the smallest noise related to the system, can detect occurred faults in the early stages.

VII. CONCLUSION

Power electronic systems (PESs) are considered to be vital components in power/energy systems, and the stability of the power/energy systems mainly depends on the health of PESs.



VI. FAULT-TOLERANT IN PESS Fault-tolerant is a prominent feature of each system so that

Level type	Method	Ref.	PES application
Switch-level	Inherently redundant switching	[195]	Three-level active NPC inverters
	states	[196]	T-Type Three-Level Inverter
		[197]	High-power active NPC three-level inverter
		[198]	Single-phase MMC-based inverter
		[199]	Interleaved DC-DC boost converters for homes and
			offices
	DC-bus midpoint connection	[200]	Five-phase permanent-magnet machine drives
		[201]	Six-phase permanent magnet synchronous motor
		[202]	A single field-programmable gate array chip
	Redundant parallel or series	[203]	Matrix converter
	switches installation	[204]	Two-level voltage source converter with IGBTs directly
			in series, Modular multi-level converter
Leg-level	Adding redundant legs in series	[205]	Open-end winding permanent magnet synchronous
	or parallel connection to main		motor system based on winding reconnection
	legs	[206]	Single dc-link dual inverters
		[207]	Three-level topology based on the NPC converter
		[208]	Three-phase permanent magnet synchronous motor
		[209]	modified topologies of NPC converter based on adding a
			fourth leg, which is based on the flying capacitor
			converter structure
Module-level	Neutral-shift	[210]	Cascaded H-bridge inverter
		[211]	MMC
		[212]	Cascaded H-bridge multilevel inverters based on level-
			shifted pulse width modulation
	DC-bus voltage reconfiguration	[213]	Modular medium-voltage drives
		[214]	Symmetrical and asymmetrical cascaded multilevel
			converters
		[215]	Static synchronous compensator based on cascaded H-
			bridge multilevel converter
	Redundant modules installation	[216]	Modular multilevel HVDC converter
		[217]	Multilevel modular capacitor-clamped DC-DC converter
		[218]	MMC

 TABLE IV

 CLASSIFICATION OF STUDIES PERFORMED IN ORDER TO FAULT-TOLERANT IN PESS BASED ON THE METHOD USED

A variety of factors cause anomalies in PESs that could threaten the system security and reliability. Such issues call for an urgent need of early fault detection in PES. In this paper, different types of faults related to PESs and various diagnostic methods were evaluated. Several literature has been reviewed since the beginning of the fault detection process in PESs. After introducing the types of faults related to PESs, a basic classification of fault detection methods based on data mining techniques and signal measurement sensors in PESs was presented and each method was introduced with a detailed description. The fault detection procedure was described by evaluating the performance of various model-based and databased methods such as artificial neural network (ANN), machine learning, and deep learning algorithms in the various studies. In a general evaluation, it was observed that modelbased techniques, due to the structural complexity and dependence on the physical model of the system, which causes significant problems in the fault detection process, are used in a limited number of cases. Among the extensive data-based techniques, deep-learning techniques due to their high capabilities in modeling the time-series state of real-time data, eliminating data-related noises which is caused by the working conditions of PES equipment, and the ideal performance in detecting the slightest correlation between the effects of the fault and detecting it, even among large volumes of data, has surpassed other data-based methods and is now commonly recommended to fault diagnosis in PESs. Given that feature extraction is considered as the key to the success of deep learning techniques in diagnostic applications, a brief description of the feature extraction process from the measured signals was presented. Finally, in a general evaluation, deep learning techniques were introduced and proposed as one of the applications of data mining for use in other areas of power electronics industry.

It should be noted that the development of power systems has led to the introduction of modern systems based on the Internet of Things and the online use of most PESs and related equipment in power systems and industrial applications. This process causes PES devices, in addition to physical faults related to the system, to be attacked by cyber-attackers, which can lead to a variety of irreparable risks. Accordingly, for future studies and to maintain the security and stability of power and energy systems, develop methods that have the ability to prognosis faults or any failure in the system and can distinguish the effects of cyber-attacks from faults. In addition, it is recommended to use deep learning-based autoencoder approaches that can reconstruct false signals from cyber-attacks in online applications and send a clean communication signal to the user or equipment.

REFERENCES

 J. Fang, H. Li, Y. Tang, and F. Blaabjerg, "Distributed Power System Virtual Inertia Implemented by Grid-Connected Power Converters," *IEEE Transactions on Power Electronics*, vol. 33, no. 10, pp. 8488– 8499, 2018, doi: 10.1109/TPEL.2017.2785218.

- [2] S. Peyghami, P. Palensky, and F. Blaabjerg, "An Overview on the Reliability of Modern Power Electronic Based Power Systems," *IEEE Open Journal of Power Electronics*, vol. 1, pp. 34–50, 2020, doi: 10.1109/ojpel.2020.2973926.
- [3] Y. Yang, A. Sangwongwanich, and F. Blaabjerg, "Design for Reliability of Power Electronics for Grid-Connected Photovoltaic Systems," *CPSS Transactions on Power Electronics and Applications*, vol. 1, no. 1, pp. 92–103, 2016, doi: 10.24295/cpsstpea.2016.00009.
- [4] Y. Song and B. Wang, "Survey on reliability of power electronic systems," *IEEE Transactions on Power Electronics*, vol. 28, no. 1, pp. 591–604, 2013, doi: 10.1109/TPEL.2012.2192503.
- [5] F. Wu, X. Li, F. Feng, and H. B. Gooi, "Multi-topology-Mode Grid-Connected Inverter to Improve Comprehensive Performance of Renewable Energy Source Generation System," *IEEE Transactions* on Power Electronics, vol. 32, no. 5, pp. 3623–3633, 2017, doi: 10.1109/TPEL.2016.2589974.
- [6] D. Zinchenko, R. Kosenko, D. Vinnikov, and A. Blinov, "Faulttolerant soft-switching current-fed DC-DC converter," in 2019 IEEE 2nd Ukraine Conference on Electrical and Computer Engineering, UKRCON 2019 - Proceedings, 2019, pp. 437–440, doi: 10.1109/UKRCON.2019.8879973.
- [7] Q. Sun, Y. Wang, and Y. Jiang, "A Novel Fault Diagnostic Approach for DC-DC Converters Based on CSA-DBN," *IEEE Access*, vol. 6, pp. 6273–6285, 2018, doi: 10.1109/ACCESS.2017.2786458.
- [8] W. Jiang, L. Huang, L. Zhang, H. Zhao, L. Wang, and W. Chen, "Control of Active Power Exchange With Auxiliary Power Loop in a Single-Phase Cascaded Multilevel Converter-Based Energy Storage System," *IEEE Transactions on Power Electronics*, vol. 32, no. 2, pp. 1518–1532, Feb. 2017, doi: 10.1109/TPEL.2016.2543751.
- [9] Y. Wu, Y. Wang, Y. Jiang, and Q. Sun, "Multiple parametric faults diagnosis for power electronic circuits based on hybrid bond graph and genetic algorithm," *Measurement: Journal of the International Measurement Confederation*, vol. 92, pp. 365–381, Oct. 2016, doi: 10.1016/j.measurement.2016.06.018.
- [10] H. Soliman, H. Wang, and F. Blaabjerg, "A Review of the Condition Monitoring of Capacitors in Power Electronic Converters," *IEEE Transactions on Industry Applications*, vol. 52, no. 6, pp. 4976–4989, Nov. 2016, doi: 10.1109/TIA.2016.2591906.
- [11] M. Pecht and R. Jaai, "A prognostics and health management roadmap for information and electronics-rich systems," *Microelectronics Reliability*, vol. 50, no. 3, pp. 317–323, Mar. 2010, doi: 10.1016/j.microrel.2010.01.006.
- [12] H. S. H. Chung, H. Wang, F. Blaabjerg, and M. Pecht, *Reliability of Power Electronic Converter Systems*. Institution of Engineering and Technology, 2015.
- [13] M. Riera-Guasp, J. A. Antonino-Daviu, and G.-A. Capolino, "Advances in Electrical Machine, Power Electronic, and Drive Condition Monitoring and Fault Detection: State of the Art," *IEEE Transactions on Industrial Electronics*, vol. 62, no. 3, pp. 1746–1759, Mar. 2015, doi: 10.1109/TIE.2014.2375853.
- [14] L. Duchesne, E. Karangelos, and L. Wehenkel, "Recent Developments in Machine Learning for Energy Systems Reliability Management," *Proceedings of the IEEE*, vol. 108, no. 9, pp. 1656– 1676, Sep. 2020, doi: 10.1109/JPROC.2020.2988715.
- [15] S. Zhao, F. Blaabjerg, and H. Wang, "An Overview of Artificial Intelligence Applications for Power Electronics," *IEEE Transactions* on *Power Electronics*, vol. 36, no. 4, pp. 4633–4658, Apr. 2021, doi: 10.1109/TPEL.2020.3024914.
- [16] B. Cai, Y. Liu, and M. Xie, "A Dynamic-Bayesian-Network-Based Fault Diagnosis Methodology Considering Transient and Intermittent Faults," *IEEE Transactions on Automation Science and Engineering*, vol. 14, no. 1, pp. 276–285, Jan. 2017, doi: 10.1109/TASE.2016.2574875.
- [17] B. Cai, H. Liu, and M. Xie, "A real-time fault diagnosis methodology of complex systems using object-oriented Bayesian networks," *Mechanical Systems and Signal Processing*, vol. 80, pp. 31–44, Dec. 2016, doi: 10.1016/j.ymssp.2016.04.019.
- [18] B. Cai *et al.*, "Fault detection and diagnostic method of diesel engine by combining rule-based algorithm and BNs/BPNNs," *Journal of Manufacturing Systems*, vol. 57, pp. 148–157, Oct. 2020, doi: 10.1016/j.jmsy.2020.09.001.
- [19] B. Cai *et al.*, "Data-driven early fault diagnostic methodology of permanent magnet synchronous motor," *Expert Systems with Applications*, vol. 177, p. 115000, Sep. 2021, doi:

10.1016/j.eswa.2021.115000.

- [20] J. Popović-Gerber, J. A. Ferreira, and J. D. Van Wyk, "Quantifying the value of power electronics in sustainable electrical energy systems," *IEEE Transactions on Power Electronics*, vol. 26, no. 12, pp. 3534–3544, 2011, doi: 10.1109/TPEL.2011.2166088.
- [21] M. R. de Gouvea, L. M. R. Baccarini, and W. M. Caminhas, "Fault detection and diagnosis in duals converters applied in DC drives," in *VII Brazilian Symposium on Neural Networks*, 2002. SBRN 2002. Proceedings., 2003, p. 110, doi: 10.1109/sbrn.2002.1181444.
- [22] N. Bawane and A. G. Kothari, "Artificial neural network based fault identification of HVDC converter," in *4th IEEE International Symposium on Diagnostics for Electric Machines, Power Electronics and Drives, 2003. SDEMPED 2003.*, 2003, pp. 152–157, doi: 10.1109/DEMPED.2003.1234564.
- [23] B. W. Jervis and J. M. Holding, "Fault diagnosis for a delta-sigma converter by a neural network," in *First International Symposium on Control, Communications and Signal Processing*, 2004., 2004, pp. 861–864, doi: 10.1109/ISCCSP.2004.1296582.
- [24] M. Andrejević and V. Litovski, "Fault diagnosis in digital part of sigma-delta converter," in 8th Seminar on Neural Network Applications in Electrical Engineering, Neurel-2006 Proceedings, 2006, pp. 177–180, doi: 10.1109/NEUREL.2006.341206.
- [25] Phan Quoc Dzung, Le Minh Phuong, Pham Quang Vinh, Nguyen Van Nho, and Dao Minh Hien, "The development of artificial neural network space vector PWM and diagnostic controller for voltage source inverter," in 2006 IEEE Power India Conference, 2006, vol. 2005, pp. 521–528, doi: 10.1109/POWERI.2006.1632564.
- [26] P. Xiao, G. K. Venayagamoorthy, K. A. Corzine, and R. Woodley, "Self-Healing Control with Multifunctional Gate Drive Circuits for Power Converters," in 2007 IEEE Industry Applications Annual Meeting, 2007, pp. 1852–1858, doi: 10.1109/07IAS.2007.282.
- [27] M. A. Masrur, H. J. Wu, C. Mi, Z. H. Chen, and Y. L. Murphey, "Fault diagnostics in power electronics-based brake-by-wire systems," *Proceedings of the Institution of Mechanical Engineers, Part D: Journal of Automobile Engineering*, vol. 222, no. 1, pp. 1– 11, Jan. 2008, doi: 10.1243/09544070JAUTO385.
- [28] J. Moshtagh, M. Jannati, H. R. Baghaee, and E. Nasr, "A novel approach for online fault detection in HVDC converters," in 2008 12th International Middle-East Power System Conference, 2008, pp. 307–311, doi: 10.1109/MEPCON.2008.4562317.
- [29] C. T. Kowalski and M. Wolkiewicz, "Stator faults diagnosis of the converter-fed induction motor using symmetrical components and neural networks," in 2009 13th European Conference on Power Electronics and Applications, EPE '09, 2009.
- [30] D. Wang and H. Lin, "Feature extraction based on supervised kernel locality preserving projection algorithm and application to fault diagnosis," in 2009 9th International Conference on Electronic Measurement & Instruments, 2009, pp. 1553–1557, doi: 10.1109/ICEMI.2009.5274804.
- [31] M. Kaminski, C. T. Kowalski, and T. Orlowska-Kowalska, "Application of radial basis neural networks for the rotor fault detection of the induction motor," in 2011 IEEE EUROCON -International Conference on Computer as a Tool, 2011, pp. 1–4, doi: 10.1109/EUROCON.2011.5929405.
- [32] R. Wang, Y. Zhan, and H. Zhou, "Application of transform in fault diagnosis of power electronics circuits," *Scientia Iranica*, vol. 19, no. 3, pp. 721–726, Jun. 2012, doi: 10.1016/j.scient.2011.06.013.
- [33] A. G. de Oliveira, R. S. T. Pontes, and C. M. de S. Medeiros, "Neural Network Used to Stator Winding Interturn Short-Circuit Fault Detection in an Induction Motor Driven by Frequency Converter," in 2013 BRICS Congress on Computational Intelligence and 11th Brazilian Congress on Computational Intelligence, 2013, pp. 459– 464, doi: 10.1109/BRICS-CCI-CBIC.2013.82.
- [34] S. Z. Lv, J. Wen, S. Y. Zhou, and W. J. Cao, "S-transform-based classification of converter faults in HVDC system by support vector machines," *Applied Mechanics and Materials*, vol. 347–350, pp. 1308–1312, Aug. 2013, doi: 10.4028/www.scientific.net/AMM.347-350.1308.
- [35] D. N. Coelho, G. A. Barreto, C. M. S. Medeiros, and J. D. A. Santos, "Performance comparison of classifiers in the detection of short circuit incipient fault in a three-phase induction motor," in 2014 IEEE Symposium on Computational Intelligence for Engineering Solutions (CIES), 2014, pp. 42–48, doi: 10.1109/CIES.2014.7011829.
- [36] P. Sanjeevikumar, B. Paily, M. Basu, and M. Conlon, "Classification of fault analysis of HVDC systems using artificial neural network,"

in 2014 49th International Universities Power Engineering Conference (UPEC), 2014, pp. 1–5, doi: 10.1109/UPEC.2014.6934775.

- [37] T. Wang, J. Qi, H. Xu, Y. Wang, L. Liu, and D. Gao, "Fault diagnosis method based on FFT-RPCA-SVM for Cascaded-Multilevel Inverter," *ISA Transactions*, vol. 60, pp. 156–163, Jan. 2016, doi: 10.1016/j.isatra.2015.11.018.
- [38] T. Wang, H. Xu, J. Han, E. Elbouchikhi, and M. E. H. Benbouzid, "Cascaded H-Bridge Multilevel Inverter System Fault Diagnosis Using a PCA and Multiclass Relevance Vector Machine Approach," *IEEE Transactions on Power Electronics*, vol. 30, no. 12, pp. 7006– 7018, Dec. 2015, doi: 10.1109/TPEL.2015.2393373.
- [39] R. B. D., "Fault Detection and Diagnosis of High Speed Switching Devices in Power Inverter," *International Journal of Research in Engineering and Technology*, vol. 04, no. 02, pp. 253–257, Feb. 2015, doi: 10.15623/ijret.2015.0402033.
- [40] J. Cui, "Faults classification of power electronic circuits based on a support vector data description method," *Metrology and Measurement Systems*, vol. 22, no. 2, pp. 205–220, Jun. 2015, doi: 10.1515/mms-2015-0017.
- [41] A. Reatti, S. Manetti, M. C. Piccirilli, A. Luchetta, and M. K. Kazimierczuk, "Multilayer Neural Network with Multivalued Neurons MLMVN based Class E resonant inverter fault detection," in *IET Conference Publications*, 2016, vol. 2016, no. CP684, pp. 6 .- 6 ., doi: 10.1049/cp.2016.0265.
- [42] C. Venkatesh and P. V. Rao, "Wavelet-ANN based classification of HVDC converter faults," in 2016 IEEE 6th International Conference on Power Systems (ICPS), 2016, pp. 1–5, doi: 10.1109/ICPES.2016.7584032.
- [43] R. B. Dhumale and S. D. Lokhande, "Neural Network Fault Diagnosis of Voltage Source Inverter under variable load conditions at different frequencies," *Measurement*, vol. 91, pp. 565–575, Sep. 2016, doi: 10.1016/j.measurement.2016.04.051.
- [44] V. Gomathy and S. Selvaperumal, "Fault Detection and Classification with Optimization Techniques for a Three-Phase Single-Inverter Circuit," *Journal of Power Electronics*, vol. 16, no. 3, pp. 1097–1109, May 2016, doi: 10.6113/JPE.2016.16.3.1097.
- [45] M. Lin, Y.-H. Li, L. Qu, C. Wu, and G.-Q. Yuan, "Fault Detection of a Proposed Three-Level Inverter Based on a Weighted Kernel Principal Component Analysis," *Journal of Power Electronics*, vol. 16, no. 1, pp. 182–189, Jan. 2016, doi: 10.6113/JPE.2016.16.1.182.
- [46] N. T. Nguyen and H. P. Nguyen, "Fault diagnosis of voltage source inverter for induction motor drives using decision tree," in *Lecture Notes in Electrical Engineering*, vol. 398, 2017, pp. 819–826.
- [47] D. N. Coelho, G. A. Barreto, and C. M. S. Medeiros, "Detection of short circuit faults in 3-phase converter-fed induction motors using kernel SOMs," in 12th International Workshop on Self-Organizing Maps and Learning Vector Quantization, Clustering and Data Visualization, WSOM 2017 - Proceedings, 2017, pp. 1–7, doi: 10.1109/WSOM.2017.8020016.
- [48] L. Shi, S. Zhang, J. Li, P. Wei, Z. Liu, and Z. Zhang, "Research on fault warning of AC filter in converter station based on RBF neural network," in 2017 6th International Conference on Computer Science and Network Technology (ICCSNT), 2017, vol. 2018-Janua, pp. 304– 309, doi: 10.1109/ICCSNT.2017.8343707.
- [49] N. Torabi, F. Naghavi, and H. A. Toliyat, "Real-time fault isolation in multiphase multilevel NPC converters using active semisupervised fuzzy clustering algorithm with pairwise constraints," in 2017 IEEE International Electric Machines and Drives Conference (IEMDC), 2017, pp. 1–7, doi: 10.1109/IEMDC.2017.8002409.
- [50] A. Sai Sarathi Vasan and M. G. Pecht, "Electronic Circuit Health Estimation Through Kernel Learning," *IEEE Transactions on Industrial Electronics*, vol. 65, no. 2, pp. 1585–1594, Feb. 2018, doi: 10.1109/TIE.2017.2733419.
- [51] H. Chen, B. Jiang, and N. Lu, "Data-Driven Incipient Sensor Fault Estimation with Application in Inverter of High-Speed Railway," *Mathematical Problems in Engineering*, vol. 2017, pp. 1–13, 2017, doi: 10.1155/2017/8937356.
- [52] B. Xin, T. Wang, and T. Tang, "A deep learning and softmax regression fault diagnosis method for multi-level converter," in *Proceedings of the 2017 IEEE 11th International Symposium on Diagnostics for Electrical Machines, Power Electronics and Drives, SDEMPED 2017*, 2017, vol. 2017-Janua, pp. 292–297, doi: 10.1109/DEMPED.2017.8062370.
- [53] L. Wang, F. Lyu, Y. Su, and J. Yue, "Kernel Entropy-Based

Classification Approach for Superbuck Converter Circuit Fault Diagnosis," *IEEE Access*, vol. 6, pp. 45504–45514, 2018, doi: 10.1109/ACCESS.2018.2864138.

- [54] M. Biglarbegian, M. Baharani, N. Kim, H. Tabkhi, and B. Parkhideh, "Scalable Reliability Monitoring of GaN Power Converter Through Recurrent Neural Networks," in 2018 IEEE Energy Conversion Congress and Exposition, ECCE 2018, 2018, pp. 7271–7277, doi: 10.1109/ECCE.2018.8557565.
- [55] S. Kiranyaz, A. Gastli, L. Ben-Brahim, N. Al-Emadi, and M. Gabbouj, "Real-Time Fault Detection and Identification for MMC Using 1-D Convolutional Neural Networks," *IEEE Transactions on Industrial Electronics*, vol. 66, no. 11, pp. 8760–8771, Nov. 2019, doi: 10.1109/TIE.2018.2833045.
- [56] Q. Lin, S. Chen, and C.-M. Lin, "Parametric Fault Diagnosis Based on Fuzzy Cerebellar Model Neural Networks," *IEEE Transactions on Industrial Electronics*, vol. 66, no. 10, pp. 8104–8115, Oct. 2019, doi: 10.1109/TIE.2018.2884195.
- [57] A. Luchetta *et al.*, "MLMVNNN for Parameter Fault Detection in PWM DC-DC Converters and Its Applications for Buck and Boost DC-DC Converters," *IEEE Transactions on Instrumentation and Measurement*, vol. 68, no. 2, pp. 439–449, Feb. 2019, doi: 10.1109/TIM.2018.2847978.
- [58] I. Martin-Diaz, D. Morinigo-Sotelo, O. Duque-Perez, and R. J. Romero-Troncoso, "An Experimental Comparative Evaluation of Machine Learning Techniques for Motor Fault Diagnosis Under Various Operating Conditions," *IEEE Transactions on Industry Applications*, vol. 54, no. 3, pp. 2215–2224, May 2018, doi: 10.1109/TIA.2018.2801863.
- [59] X. Qu, B. Duan, Q. Yin, M. Shen, and Y. Yan, "Deep Convolution Neural Network Based Fault Detection and Identification for Modular Multilevel Converters," in 2018 IEEE Power & Energy Society General Meeting (PESGM), 2018, vol. 2018-Augus, pp. 1–5, doi: 10.1109/PESGM.2018.8586661.
- [60] C. Li, Z. Liu, Y. Zhang, L. Chai, and B. Xu, "Diagnosis and location of the open-circuit fault in modular multilevel converters: An improved machine learning method," *Neurocomputing*, vol. 331, pp. 58–66, Feb. 2019, doi: 10.1016/j.neucom.2018.09.041.
- [61] M. Fei, L. Ning, M. Huiyu, P. Yi, S. Haoyuan, and Z. Jianyong, "Online fault diagnosis model for locomotive traction inverter based on wavelet transform and support vector machine," *Microelectronics Reliability*, vol. 88–90, pp. 1274–1280, Sep. 2018, doi: 10.1016/j.microrel.2018.06.069.
- [62] L. Yi, X. Ren, H. Zhu, and B. Wang, "Fault Diagnosis of Phase Shifted Full Bridge Converter Based on Wavelet Packet and Extreme Learning Machine Optimized by Firefly Algorithm," *DEStech Transactions on Environment, Energy and Earth Sciences*, no. appeec, Jun. 2018, doi: 10.12783/dtees/appeec2018/23503.
- [63] L. Xu, M. Cao, B. Song, J. Zhang, Y. Liu, and F. E. Alsaadi, "Opencircuit fault diagnosis of power rectifier using sparse autoencoder based deep neural network," *Neurocomputing*, vol. 311, pp. 1–10, Oct. 2018, doi: 10.1016/j.neucom.2018.05.040.
- [64] Y. Tan, H. Zhang, and Y. Zhou, "Fault Detection Method for Permanent Magnet Synchronous Generator Wind Energy Converters Using Correlation Features Among Three-phase Currents," *Journal* of Modern Power Systems and Clean Energy, vol. 8, no. 1, pp. 168– 178, 2020, doi: 10.35833/MPCE.2018.000347.
- [65] Z. Ke *et al.*, "Single-submodule open-circuit fault diagnosis for a modular multi-level converter using artificial intelligent-based techniques," in *Conference Proceedings - IEEE Applied Power Electronics Conference and Exposition - APEC*, 2019, vol. 2019-March, pp. 3056–3063, doi: 10.1109/APEC.2019.8721984.
- [66] Z. Li et al., "A Fast Diagnosis Method for Both IGBT Faults and Current Sensor Faults in Grid-Tied Three-Phase Inverters With Two Current Sensors," *IEEE Transactions on Power Electronics*, vol. 35, no. 5, pp. 5267–5278, May 2020, doi: 10.1109/TPEL.2019.2946692.
- [67] S. Ye, J. Jiang, J. Li, Y. Liu, Z. Zhou, and C. Liu, "Fault Diagnosis and Tolerance Control of Five-Level Nested NPP Converter Using Wavelet Packet and LSTM," *IEEE Transactions on Power Electronics*, vol. 35, no. 2, pp. 1907–1921, Feb. 2020, doi: 10.1109/TPEL.2019.2921677.
- [68] Z. Y. Xue, M. S. Li, K. S. Xiahou, T. Y. Ji, and Q. H. Wu, "A Data-Driven Diagnosis Method of Open-Circuit Switch Faults for PMSG-Based Wind Generation System," in *Proceedings of the 2019 IEEE* 12th International Symposium on Diagnostics for Electrical Machines, Power Electronics and Drives, SDEMPED 2019, 2019,

pp. 481-487, doi: 10.1109/DEMPED.2019.8864922.

- [69] M. Manohar, E. Koley, and S. Ghosh, "Enhancing resilience of PVfed microgrid by improved relaying and differentiating between inverter faults and distribution line faults," *International Journal of Electrical Power & Energy Systems*, vol. 108, pp. 271–279, Jun. 2019, doi: 10.1016/j.ijepes.2019.01.015.
- [70] S. H. Kim, D. Y. Yoo, S. W. An, Y. S. Park, J. W. Lee, and K. B. Lee, "Fault Detection Method Using a Convolution Neural Network for Hybrid Active Neutral-Point Clamped Inverters," *IEEE Access*, vol. 8, pp. 140632–140642, 2020, doi: 10.1109/ACCESS.2020.3011730.
- [71] Y. Xia, Y. Xu, and B. Gou, "A Data-Driven Method for IGBT Open-Circuit Fault Diagnosis Based on Hybrid Ensemble Learning and Sliding-Window Classification," *IEEE Transactions on Industrial Informatics*, vol. 16, no. 8, pp. 5223–5233, Aug. 2020, doi: 10.1109/TII.2019.2949344.
- [72] H. Dong, F. Chen, zhipeng wang, L. Jia, Y. Qin, and J. Man, "An Adaptive Multi-Sensor Fault Diagnosis Method for High-Speed Train Traction Converters," *IEEE Transactions on Power Electronics*, vol. 36, no. 6, pp. 1–1, Jun. 2020, doi: 10.1109/tpel.2020.3034190.
- [73] W. Chen, L. Zhang, K. Pattipati, A. M. Bazzi, S. Joshi, and E. M. Dede, "Data-Driven Approach for Fault Prognosis of SiC MOSFETs," *IEEE Transactions on Power Electronics*, vol. 35, no. 4, pp. 4048–4062, Apr. 2020, doi: 10.1109/TPEL.2019.2936850.
- [74] G. Yating, W. Wu, L. Qiongbin, C. Fenghuang, and C. Qinqin, "Fault Diagnosis for Power Converters Based on Optimized Temporal Convolutional Network," *IEEE Transactions on Instrumentation and Measurement*, vol. 70, pp. 1–10, 2021, doi: 10.1109/TIM.2020.3021110.
- [75] T. Kim, H.-W. Lee, and S. Kwak, "Open-Circuit Switch-Fault Tolerant Control of a Modified Boost DC–DC Converter for Alternative Energy Systems," *IEEE Access*, vol. 7, pp. 69535–69544, 2019, doi: 10.1109/ACCESS.2019.2919238.
- [76] G. W. Koo, B. K. Lee, and D. Joo, "Threshold Point Calculation Method Using Instantaneous Phase Current for Switch Fault Diagnosis of AC-DC Converters in Hybrid Grid Systems," in 2019 IEEE Applied Power Electronics Conference and Exposition (APEC), 2019, vol. 2019-March, pp. 1065–1069, doi: 10.1109/APEC.2019.8721962.
- [77] L. M. Halabi, I. M. Alsofyani, and K. B. Lee, "Multiple-fault-Tolerant strategy for three-phase hybrid active neutral point clamped converters using enhanced space vector modulation technique," *IEEE Access*, vol. 8, pp. 180113–180123, 2020, doi: 10.1109/ACCESS.2020.3028115.
- [78] C. Yang *et al.*, "Voltage Difference Residual-Based Open-Circuit Fault Diagnosis Approach for Three-Level Converters in Electric Traction Systems," *IEEE Transactions on Power Electronics*, vol. 35, no. 3, pp. 3012–3028, Mar. 2020, doi: 10.1109/TPEL.2019.2924487.
- [79] C. Liu *et al.*, "Fault Localization Strategy for Modular Multilevel Converters under Submodule Lower Switch Open-Circuit Fault," *IEEE Transactions on Power Electronics*, vol. 35, no. 5, pp. 5190– 5204, May 2020, doi: 10.1109/TPEL.2019.2941129.
- [80] A. Kiselev, G. R. Catuogno, A. Kuznietsov, and R. Leidhold, "Finite-Control-Set MPC for Open-Phase Fault-Tolerant Control of PM Synchronous Motor Drives," *IEEE Transactions on Industrial Electronics*, vol. 67, no. 6, pp. 4444–4452, Jun. 2020, doi: 10.1109/TIE.2019.2931285.
- [81] H. Mhiesan, Y. Wei, Y. P. Siwakoti, and H. A. Mantooth, "A Fault-Tolerant Hybrid Cascaded H-Bridge Multilevel Inverter," *IEEE Transactions on Power Electronics*, vol. 35, no. 12, pp. 12702– 12715, Dec. 2020, doi: 10.1109/TPEL.2020.2996097.
- [82] X. Guo, S. Sui, B. Wang, and W. Zhang, "A current-based approach for short-circuit fault diagnosis in closed-loop current source inverter," *IEEE Transactions on Industrial Electronics*, vol. 67, no. 9, pp. 7941–7950, Sep. 2020, doi: 10.1109/TIE.2019.2941143.
- [83] J. de Kleer and J. Kurien, "Fundamentals of model-based diagnosis," *IFAC Proceedings Volumes*, vol. 36, no. 5, pp. 25–36, Jun. 2003, doi: 10.1016/S1474-6670(17)36467-4.
- [84] M. Alavi, M. Luo, D. Wang, and Danhong Zhang, "Fault diagnosis for power electronic inverters: A model-based approach," in 8th IEEE Symposium on Diagnostics for Electrical Machines, Power Electronics & Drives, 2011, pp. 221–228, doi: 10.1109/DEMPED.2011.6063627.
- [85] M. Yu, D. Wang, and Lei Huang, "Incipient fault diagnosis and

prognosis for hybrid systems with unknown mode changes," in 2010 Prognostics and System Health Management Conference, 2010, pp. 1–7, doi: 10.1109/PHM.2010.5413418.

- [86] Chang Boon Low, Danwei Wang, S. Arogeti, and Ming Luo, "Quantitative Hybrid Bond Graph-Based Fault Detection and Isolation," *IEEE Transactions on Automation Science and Engineering*, vol. 7, no. 3, pp. 558–569, Jul. 2010, doi: 10.1109/TASE.2009.2024538.
- [87] A. Pyrkin, R. Cisneros, D. U. Campos-Delgado, A. Bobtsov, and S. Somov, "A model-based fault-detection strategy in DC/AC conversion," *IFAC-PapersOnLine*, vol. 53, no. 2, pp. 676–681, 2020, doi: 10.1016/j.ifacol.2020.12.814.
- [88] Z. Li, Y. Gao, X. Zhang, B. Wang, and H. Ma, "A Model-Data-Hybrid-Driven Diagnosis Method for Open-Switch Faults in Power Converters," *IEEE Transactions on Power Electronics*, vol. 36, no. 5, pp. 4965–4970, May 2021, doi: 10.1109/TPEL.2020.3026176.
- [89] J. Poon, P. Jain, I. C. Konstantakopoulos, C. Spanos, S. K. Panda, and S. R. Sanders, "Model-based fault detection and identification for switching power converters," *IEEE Transactions on Power Electronics*, vol. 32, no. 2, pp. 1419–1430, Feb. 2017, doi: 10.1109/TPEL.2016.2541342.
- [90] X. Ge, J. Pu, B. Gou, and Y.-C. Liu, "An Open-Circuit Fault Diagnosis Approach for Single-Phase Three-Level Neutral-Point-Clamped Converters," *IEEE Transactions on Power Electronics*, vol. 33, no. 3, pp. 2559–2570, Mar. 2018, doi: 10.1109/TPEL.2017.2691804.
- [91] B. Gou, X. Ge, S. Wang, X. Feng, J. B. Kuo, and T. G. Habetler, "An Open-Switch Fault Diagnosis Method for Single-Phase PWM Rectifier Using a Model-Based Approach in High-Speed Railway Electrical Traction Drive System," *IEEE Transactions on Power Electronics*, vol. 31, no. 5, pp. 3816–3826, May 2016, doi: 10.1109/TPEL.2015.2465299.
- [92] D. Gonzalez-Jimenez, J. Del-Olmo, J. Poza, F. Garramiola, and P. Madina, "Data-Driven Fault Diagnosis for Electric Drives: A Review," *Sensors*, vol. 21, no. 12, p. 4024, Jun. 2021, doi: 10.3390/s21124024.
- [93] A. K. Jain, J. Mao, and K. M. Mohiuddin, "Artificial neural networks: A tutorial," *Computer*, vol. 29, no. 3, pp. 31–44, Mar. 1996, doi: 10.1109/2.485891.
- [94] T. Kohonen, "An introduction to neural computing," *Neural Networks*, vol. 1, no. 1, pp. 3–16, Jan. 1988, doi: 10.1016/0893-6080(88)90020-2.
- [95] P. Buduma, N. K. Vulisi, and G. Panda, "Robust Control and Kalman MPPT for Grid Assimilated Wind Energy Conversion System," *IEEE Transactions on Industry Applications*, pp. 1–1, 2020, doi: 10.1109/TIA.2020.3047585.
- [96] X. Fu and S. Li, "Control of Single-Phase Grid-Connected Converters with LCL Filters Using Recurrent Neural Network and Conventional Control Methods," *IEEE Transactions on Power Electronics*, vol. 31, no. 7, pp. 5354–5364, 2016, doi: 10.1109/TPEL.2015.2490200.
- [97] R.-J. Wai, M.-W. Chen, and Y.-K. Liu, "Design of Adaptive Control and Fuzzy Neural Network Control for Single-Stage Boost Inverter," *IEEE Transactions on Industrial Electronics*, vol. 62, no. 9, pp. 5434–5445, Sep. 2015, doi: 10.1109/TIE.2015.2408571.
- [98] S. Li, M. Fairbank, C. Johnson, D. C. Wunsch, E. Alonso, and J. L. Proao, "Artificial neural networks for control of a grid-Connected rectifier/inverter under disturbance, dynamic and power converter switching conditions," *IEEE Transactions on Neural Networks and Learning Systems*, vol. 25, no. 4, pp. 738–750, Apr. 2014, doi: 10.1109/TNNLS.2013.2280906.
- [99] I. S. Mohamed, S. Rovetta, T. D. Do, T. Dragicevic, and A. A. Z. Diab, "A Neural-Network-Based Model Predictive Control of Three-Phase Inverter With an Output \$LC\$ Filter," *IEEE Access*, vol. 7, pp. 124737–124749, 2019, doi: 10.1109/ACCESS.2019.2938220.
- [100] X. Zhan, W. Wang, and H. Chung, "A Neural-Network-Based Color Control Method for Multi-Color LED Systems," *IEEE Transactions* on *Power Electronics*, vol. 34, no. 8, pp. 7900–7913, Aug. 2019, doi: 10.1109/TPEL.2018.2880876.
- [101] D. Tran and Y. K. Tan, "Sensorless Illumination Control of a Networked LED-Lighting System Using Feedforward Neural Network," *IEEE Transactions on Industrial Electronics*, vol. 61, no. 4, pp. 2113–2121, Apr. 2014, doi: 10.1109/TIE.2013.2266084.
- [102] A. El Mejdoubi, H. Chaoui, J. Sabor, and H. Gualous, "Remaining Useful Life Prognosis of Supercapacitors Under Temperature and Voltage Aging Conditions," *IEEE Transactions on Industrial*

Electronics, vol. 65, no. 5, pp. 4357–4367, May 2018, doi: 10.1109/TIE.2017.2767550.

- [103] C. S. Fahn, K. T. Lan, and Z. B. Chern, "Fuzzy rules generation using new evolutionary algorithms combined with multilayer perceptrons," *IEEE Transactions on Industrial Electronics*, vol. 46, no. 6, pp. 1103–1113, 1999, doi: 10.1109/41.807995.
- [104] A. Moradzadeh, A. Mansour-Saatloo, B. Mohammadi-Ivatloo, and A. Anvari-Moghaddam, "Performance evaluation of two machine learning techniques in heating and cooling loads forecasting of residential buildings," *Applied Sciences (Switzerland)*, vol. 10, no. 11, p. 3829, May 2020, doi: 10.3390/app10113829.
- [105] C. K. Ku, C. H. Goay, N. S. Ahmad, and P. Goh, "Jitter Decomposition of High-Speed Data Signals From Jitter Histograms With a Pole–Residue Representation Using Multilayer Perceptron Neural Networks," *IEEE Transactions on Electromagnetic Compatibility*, vol. 62, no. 5, pp. 2227–2237, Oct. 2020, doi: 10.1109/TEMC.2019.2936000.
- [106] Y. Fan, B. Tao, Y. Zheng, and S. S. Jang, "A Data-Driven Soft Sensor Based on Multilayer Perceptron Neural Network with a Double LASSO Approach," *IEEE Transactions on Instrumentation and Measurement*, vol. 69, no. 7, pp. 3972–3979, Jul. 2020, doi: 10.1109/TIM.2019.2947126.
- [107] T. Kohonen, "Self-organized formation of topologically correct feature maps," *Biological Cybernetics*, vol. 43, no. 1, pp. 59–69, 1982, doi: 10.1007/BF00337288.
- [108] O. Linda and M. Manic, "Self-Organizing Fuzzy Haptic Teleoperation of Mobile Robot Using Sparse Sonar Data," *IEEE Transactions on Industrial Electronics*, vol. 58, no. 8, pp. 3187–3195, Aug. 2011, doi: 10.1109/TIE.2009.2037649.
- [109] X. Li and D. Zhu, "An Adaptive SOM Neural Network Method to Distributed Formation Control of a Group of AUVs," *IEEE Transactions on Industrial Electronics*, vol. 65, no. 10, pp. 8260– 8270, 2018, doi: 10.1109/TIE.2018.2807368.
- [110] Q. Zhou *et al.*, "An active learning radial basis function modeling method based on self-organization maps for simulation-based design problems," *Knowledge-Based Systems*, vol. 131, pp. 10–27, Sep. 2017, doi: 10.1016/j.knosys.2017.05.025.
- [111] J.-S. R. Jang, "ANFIS: adaptive-network-based fuzzy inference system," *IEEE Transactions on Systems, Man, and Cybernetics*, vol. 23, no. 3, pp. 665–685, 1993, doi: 10.1109/21.256541.
- [112] M. Zeinalnezhad, A. G. Chofreh, F. A. Goni, and J. J. Klemeš, "Air pollution prediction using semi-experimental regression model and Adaptive Neuro-Fuzzy Inference System," *Journal of Cleaner Production*, vol. 261, p. 121218, Jul. 2020, doi: 10.1016/j.jclepro.2020.121218.
- [113] M. Singh and A. Chandra, "Application of Adaptive Network-Based Fuzzy Inference System for Sensorless Control of PMSG-Based Wind Turbine With Nonlinear-Load-Compensation Capabilities," *IEEE Transactions on Power Electronics*, vol. 26, no. 1, pp. 165–175, Jan. 2011, doi: 10.1109/TPEL.2010.2054113.
- [114] M. Vakili, M. Yahyaei, J. Ramsay, P. Aghajannezhad, and B. Paknezhad, "Adaptive neuro-fuzzy inference system modeling to predict the performance of graphene nanoplatelets nanofluid-based direct absorption solar collector based on experimental study," *Renewable Energy*, vol. 163, pp. 807–824, Jan. 2021, doi: 10.1016/j.renene.2020.08.134.
- [115] Q. T. Tran, S. D. Nguyen, and T.-I. Seo, "Algorithm for Estimating Online Bearing Fault Upon the Ability to Extract Meaningful Information From Big Data of Intelligent Structures," *IEEE Transactions on Industrial Electronics*, vol. 66, no. 5, pp. 3804–3813, May 2019, doi: 10.1109/TIE.2018.2847704.
- [116] A. Fotouhi, D. J. Auger, K. Propp, and S. Longo, "Lithium–Sulfur Battery State-of-Charge Observability Analysis and Estimation," *IEEE Transactions on Power Electronics*, vol. 33, no. 7, pp. 5847– 5859, Jul. 2018, doi: 10.1109/TPEL.2017.2740223.
- [117] H. R. Baghaee, M. Mirsalim, G. B. Gharehpetian, H. A. Talebi, and A. Niknam-Kumle, "Notice of Violation of IEEE Publication Principles: A Hybrid ANFIS/ABC-based Online Selective Harmonic Elimination Switching Pattern for Cascaded Multi-level Inverters of Microgrids," *IEEE Transactions on Industrial Electronics*, pp. 1–1, 2017, doi: 10.1109/tie.2017.2694403.
- [118] D. BROOMHEAD, "Multivariable functional interpolation and adaptive networks," *Complex Systems*, vol. 2, pp. 321–355, 1988.
- [119] L. Liang, W. Guo, Y. Zhang, W. Z. Zhang, L. Li, and X. Xing, "Radial Basis Function Neural Network for prediction of medium-

frequency sound absorption coefficient of composite structure opencell aluminum foam," *Applied Acoustics*, vol. 170, p. 107505, Dec. 2020, doi: 10.1016/j.apacoust.2020.107505.

- [120] D. Karamichailidou, V. Kaloutsa, and A. Alexandridis, "Wind turbine power curve modeling using radial basis function neural networks and tabu search," *Renewable Energy*, vol. 163, pp. 2137– 2152, Jan. 2021, doi: 10.1016/j.renene.2020.10.020.
- [121] G. Y. Lian, K. L. Huang, J. H. Chen, and F. Q. Gao, "Training algorithm for radial basis function neural network based on quantumbehaved particle swarm optimization," *International Journal of Computer Mathematics*, vol. 87, no. 3, pp. 629–641, Mar. 2010, doi: 10.1080/00207160802166465.
- [122] M. Babaie, M. Mehrasa, M. Sharifzadeh, and K. Al-Haddad, "Floating Weighting Factors ANN-MPC Based on Lyapunov Stability for Seven-Level Modified PUC Active Rectifier," *IEEE Transactions on Industrial Electronics*, pp. 1–1, 2021, doi: 10.1109/TIE.2021.3050375.
- [123] S. Xie, Y. Xie, T. Huang, W. Gui, and C. Yang, "Generalized Predictive Control for Industrial Processes Based on Neuron Adaptive Splitting and Merging RBF Neural Network," *IEEE Transactions on Industrial Electronics*, vol. 66, no. 2, pp. 1192–1202, Feb. 2019, doi: 10.1109/TIE.2018.2835402.
- [124] Y. Yin et al., "Observer-Based Adaptive Sliding Mode Control of NPC Converters: An RBF Neural Network Approach," *IEEE Transactions on Power Electronics*, vol. 34, no. 4, pp. 3831–3841, Apr. 2019, doi: 10.1109/TPEL.2018.2853093.
- [125] W. M. Lin, C. M. Hong, and F. S. Cheng, "Fuzzy neural network output maximization control for sensorless wind energy conversion system," *Energy*, vol. 35, no. 2, pp. 592–601, Feb. 2010, doi: 10.1016/j.energy.2009.10.030.
- [126] K. Sabahi, M. Teshnehlab, and M. A. Shoorhedeli, "Recurrent fuzzy neural network by using feedback error learning approaches for LFC in interconnected power system," *Energy Conversion and Management*, vol. 50, no. 4, pp. 938–946, Apr. 2009, doi: 10.1016/j.enconman.2008.12.028.
- [127] Y.-C. Hung, F.-J. Lin, J.-C. Hwang, J.-K. Chang, and K.-C. Ruan, "Wavelet Fuzzy Neural Network With Asymmetric Membership Function Controller for Electric Power Steering System via Improved Differential Evolution," *IEEE Transactions on Power Electronics*, vol. 30, no. 4, pp. 2350–2362, Apr. 2015, doi: 10.1109/TPEL.2014.2327693.
- [128] K. H. Tan, F. J. Lin, C. M. Shih, and C. N. Kuo, "Intelligent Control of Microgrid with Virtual Inertia Using Recurrent Probabilistic Wavelet Fuzzy Neural Network," *IEEE Transactions on Power Electronics*, vol. 35, no. 7, pp. 7451–7464, Jul. 2020, doi: 10.1109/TPEL.2019.2954740.
- [129] V. N. Vapnik, *The Nature of Statistical Learning Theory*. New York, NY: Springer New York, 1995.
- [130] A. Moradzadeh, S. Zakeri, M. Shoaran, B. Mohammadi-Ivatloo, and F. Mohammadi, "Short-term load forecasting of microgrid via hybrid support vector regression and long short-term memory algorithms," *Sustainability (Switzerland)*, vol. 12, no. 17, p. 7076, Aug. 2020, doi: 10.3390/su12177076.
- [131] J. C. Alvarez Anton, P. J. Garcia Nieto, C. Blanco Viejo, and J. A. Vilan Vilan, "Support vector machines used to estimate the battery state of charge," *IEEE Transactions on Power Electronics*, vol. 28, no. 12, pp. 5919–5926, Dec. 2013, doi: 10.1109/TPEL.2013.2243918.
- [132] A. Moradzadeh, S. Zeinal-Kheiri, B. Mohammadi-Ivatloo, M. Abapour, and A. Anvari-Moghaddam, "Support Vector Machine-Assisted Improvement Residential Load Disaggregation," in 2020 28th Iranian Conference on Electrical Engineering (ICEE), 2020, pp. 1–6, doi: 10.1109/ICEE50131.2020.9260869.
- [133] J. Meng, G. Luo, and F. Gao, "Lithium polymer battery state-ofcharge estimation based on adaptive unscented kalman filter and support vector machine," *IEEE Transactions on Power Electronics*, vol. 31, no. 3, pp. 2226–2238, Mar. 2016, doi: 10.1109/TPEL.2015.2439578.
- [134] M. Kang, J. Kim, J.-M. Kim, A. C. C. Tan, E. Y. Kim, and B.-K. Choi, "Reliable Fault Diagnosis for Low-Speed Bearings Using Individually Trained Support Vector Machines With Kernel Discriminative Feature Analysis," *IEEE Transactions on Power Electronics*, vol. 30, no. 5, pp. 2786–2797, May 2015, doi: 10.1109/TPEL.2014.2358494.
- [135] L. Cai, J. Meng, D.-I. Stroe, J. Peng, G. Luo, and R. Teodorescu,

"Multiobjective Optimization of Data-Driven Model for Lithium-Ion Battery SOH Estimation With Short-Term Feature," *IEEE Transactions on Power Electronics*, vol. 35, no. 11, pp. 11855– 11864, Nov. 2020, doi: 10.1109/TPEL.2020.2987383.

- [136] A. Chakrabarty, G. T. Buzzard, and S. H. Zak, "Output-Tracking Quantized Explicit Nonlinear Model Predictive Control Using Multiclass Support Vector Machines," *IEEE Transactions on Industrial Electronics*, vol. 64, no. 5, pp. 4130–4138, May 2017, doi: 10.1109/TIE.2016.2638401.
- [137] G.-B. Huang, Q.-Y. Zhu, and C.-K. Siew, "Extreme learning machine: theory and applications," *Neurocomputing*, vol. 70, no. 1– 3, pp. 489–501, 2006.
- [138] C. Liu, B. Sun, C. Zhang, and F. Li, "A hybrid prediction model for residential electricity consumption using holt-winters and extreme learning machine," *Applied Energy*, vol. 275, p. 115383, Oct. 2020, doi: 10.1016/j.apenergy.2020.115383.
- [139] L. Wang, X. Li, and Y. Bai, "Short-term wind speed prediction using an extreme learning machine model with error correction," *Energy Conversion and Management*, vol. 162, pp. 239–250, Apr. 2018, doi: 10.1016/j.enconman.2018.02.015.
- [140] J.-M. Huang, R.-J. Wai, and G.-J. Yang, "Design of Hybrid Artificial Bee Colony Algorithm and Semi-Supervised Extreme Learning Machine for PV Fault Diagnoses by Considering Dust Impact," *IEEE Transactions on Power Electronics*, vol. 35, no. 7, pp. 7086–7099, Jul. 2020, doi: 10.1109/TPEL.2019.2956812.
- [141] C. M. Bishop, *Bishop Pattern Recognition And Machine Learning -*Springer 2006. 2014.
- [142] D. Wu, H. Chen, Y. Huang, and S. Chen, "Online Monitoring and Model-Free Adaptive Control of Weld Penetration in VPPAW Based on Extreme Learning Machine," *IEEE Transactions on Industrial Informatics*, vol. 15, no. 5, pp. 2732–2740, May 2019, doi: 10.1109/TII.2018.2870933.
- [143] M. S. Hossain Lipu, M. A. Hannan, A. Hussain, M. H. Saad, A. Ayob, and M. N. Uddin, "Extreme learning machine model for state-ofcharge estimation of lithium-ion battery using gravitational search algorithm," *IEEE Transactions on Industry Applications*, vol. 55, no. 4, pp. 4225–4234, 2019, doi: 10.1109/TIA.2019.2902532.
- [144] J. Song, F. Dong, J. Zhao, H. Wang, Z. He, and L. Wang, "An Efficient Multiobjective Design Optimization Method for a PMSLM Based on an Extreme Learning Machine," *IEEE Transactions on Industrial Electronics*, vol. 66, no. 2, pp. 1001–1011, Feb. 2019, doi: 10.1109/TIE.2018.2835413.
- [145] S. F. Stefenon, R. B. Grebogi, R. Z. Freire, A. Nied, and L. H. Meyer, "Optimized Ensemble Extreme Learning Machine for Classification of Electrical Insulators Conditions," *IEEE Transactions on Industrial Electronics*, vol. 67, no. 6, pp. 5170–5178, Jun. 2020, doi: 10.1109/TIE.2019.2926044.
- [146] D. Gao, Q. Guo, M. Jin, Y. Yu, and J. Xi, "Adaptive Extreme Learning Machine-Based Nonlinearity Mitigation For LED Communications," *IEEE Journal of Selected Topics in Quantum Electronics*, vol. 27, no. 2, pp. 1–9, Mar. 2021, doi: 10.1109/JSTQE.2020.3043779.
- [147] Y. Xu, Z. Zhang, Z. Bian, and L. Yu, "Copper Loss Optimization Based on Bidirectional Converter for Doubly Salient Brushless Starter/Generator System," *IEEE Transactions on Industrial Electronics*, pp. 1–1, 2020, doi: 10.1109/TIE.2020.2988239.
- [148] J.-M. Kim, K.-B. Lee, D.-C. Lee, and Y.-J. Ko, "Fault diagnosis of three-parallel voltage-source converter for a high-power wind turbine," *IET Power Electronics*, vol. 5, no. 7, pp. 1058–1067, Aug. 2012, doi: 10.1049/iet-pel.2011.0109.
- [149] A. Mohammadi, D. Guilbert, A. Gaillard, D. Bouquain, D. Khaburi, and A. Djerdir, "Faults diagnosis between PEM fuel cell and DC/DC converter using neural networks for automotive applications," in *IECON Proceedings (Industrial Electronics Conference)*, 2013, pp. 8186–8191, doi: 10.1109/IECON.2013.6700503.
- [150] S. S. Moosavi, A. Djerdir, Y. Ait-Amirat, D. A. Khaburi, and A. N'Diaye, "Artificial neural network-based fault diagnosis in the AC-DC converter of the power supply of series hybrid electric vehicle," *IET Electrical Systems in Transportation*, vol. 6, no. 2, pp. 96–106, Jun. 2016, doi: 10.1049/iet-est.2014.0055.
- [151] G. H. Bazan, P. R. Scalassara, W. Endo, A. Goedtel, R. H. C. Palacios, and W. F. Godoy, "Stator Short-Circuit Diagnosis in Induction Motors Using Mutual Information and Intelligent Systems," *IEEE Transactions on Industrial Electronics*, vol. 66, no. 4, pp. 3237–3246, Apr. 2019, doi: 10.1109/TIE.2018.2840983.

- [152] J. Cui, G. Shi, and C. Gong, "A fast classification method of faults in power electronic circuits based on support vector machines," *Metrology and Measurement Systems*, vol. 24, no. 4, pp. 701–720, 2017, doi: 10.1515/mms-2017-0056.
- [153] Z. Y. Xue, K. S. Xiahou, M. S. Li, T. Y. Ji, and Q. H. Wu, "Diagnosis of Multiple Open-Circuit Switch Faults Based on Long Short-Term Memory Network for DFIG-Based Wind Turbine Systems," *IEEE Journal of Emerging and Selected Topics in Power Electronics*, vol. 8, no. 3, pp. 2600–2610, Sep. 2020, doi: 10.1109/JESTPE.2019.2908981.
- [154] Z. Leonowicz and T. Lobos, "Application of time-frequency distribution and neural networks for fault classification in power electronics," in *IEEE International Conference on Computational Intelligence for Measurement Systems and Applications Proceedings*, 2003, vol. 2003-Janua, pp. 67–71, doi: 10.1109/CIMSA.2003.1227204.
- [155] P. Nagaraja and G. Sadashivappa, "Fault diagnosis with statistical properties and implementation using MKSVM for flash ADC," in *Proceedings of the 2017 IEEE International Conference on Communication and Signal Processing, ICCSP 2017*, 2018, vol. 2018-Janua, pp. 210–215, doi: 10.1109/ICCSP.2017.8286822.
- [156] A. Ahmadi, M. Nabipour, B. Mohammadi-Ivatloo, A. M. Amani, S. Rho, and M. J. Piran, "Long-Term Wind Power Forecasting Using Tree-Based Learning Algorithms," *IEEE Access*, vol. 8, pp. 151511– 151522, 2020, doi: 10.1109/ACCESS.2020.3017442.
- [157] F. Li *et al.*, "Detection and Diagnosis of Data Integrity Attacks in Solar Farms Based on Multilayer Long Short-Term Memory Network," *IEEE Transactions on Power Electronics*, vol. 36, no. 3, pp. 2495–2498, Mar. 2021, doi: 10.1109/TPEL.2020.3017935.
- [158] L. Vanfretti and V. S. N. Arava, "Decision tree-based classification of multiple operating conditions for power system voltage stability assessment," *International Journal of Electrical Power & Energy Systems*, vol. 123, p. 106251, Dec. 2020, doi: 10.1016/j.ijepes.2020.106251.
- [159] M. Namazkhan, C. Albers, and L. Steg, "A decision tree method for explaining household gas consumption: The role of building characteristics, socio-demographic variables, psychological factors and household behaviour," *Renewable and Sustainable Energy Reviews*, vol. 119, p. 109542, Mar. 2020, doi: 10.1016/j.rser.2019.109542.
- [160] L. Wang and Z. Zhang, "Automatic Detection of Wind Turbine Blade Surface Cracks Based on UAV-Taken Images," *IEEE Transactions* on Industrial Electronics, vol. 64, no. 9, pp. 7293–7309, Sep. 2017, doi: 10.1109/TIE.2017.2682037.
- [161] Y. Zheng, Y. Lu, W. Gao, X. Han, X. Feng, and M. Ouyang, "Micro-Short -Circuit Cell Fault Identification Method for Lithium-ion Battery Packs Based on the Mutual Information," *IEEE Transactions* on *Industrial Electronics*, pp. 1–1, 2020, doi: 10.1109/tie.2020.2984441.
- [162] L. JIN, F. CHEN, C. TANG, and B. JIANG, "Multiple Fault Diagnosis for Three-level Inverter Via Wavelet Analysis and Support Vector Machine," *DEStech Transactions on Engineering and Technology Research*, no. icca, Mar. 2017, doi: 10.12783/dtetr/icca2016/6009.
- [163] Y. Jiang, Y. Wang, Q. Sun, and Y. Wu, "A fault prediction method for closed-loop SEPIC converters under variable operating conditions," in *Proceedings of 2016 Prognostics and System Health Management Conference, PHM-Chengdu 2016*, 2017, pp. 1–7, doi: 10.1109/PHM.2016.7819816.
- [164] Y. Lecun, Y. Bengio, and G. Hinton, "Deep learning," *Nature*, vol. 521, no. 7553. pp. 436–444, 2015, doi: 10.1038/nature14539.
- [165] M. Sahani and P. K. Dash, "FPGA-Based Deep Convolutional Neural Network of Process Adaptive VMD Data with Online Sequential RVFLN for Power Quality Events Recognition," *IEEE Transactions* on Power Electronics, vol. 36, no. 4, pp. 4006–4015, Apr. 2021, doi: 10.1109/TPEL.2020.3023770.
- [166] A. Moradzadeh, B. Mohammadi-Ivatloo, M. Abapour, A. Anvari-Moghaddam, S. Gholami Farkoush, and S.-B. Rhee, "A practical solution based on convolutional neural network for non-intrusive load monitoring," *Journal of Ambient Intelligence and Humanized Computing*, Jan. 2021, doi: 10.1007/s12652-020-02720-6.
- [167] W. Yu and C. Zhao, "Broad Convolutional Neural Network Based Industrial Process Fault Diagnosis with Incremental Learning Capability," *IEEE Transactions on Industrial Electronics*, vol. 67, no. 6, pp. 5081–5091, Jun. 2020, doi: 10.1109/TIE.2019.2931255.

- [168] H. Teimourzadeh, A. Moradzadeh, M. Shoaran, B. Mohammadi-Ivatloo, and R. Razzaghi, "High Impedance Single-Phase Faults Diagnosis in Transmission Lines via Deep Reinforcement Learning of Transfer Functions," *IEEE Access*, vol. 9, pp. 15796–15809, 2021, doi: 10.1109/ACCESS.2021.3051411.
- [169] L. Wen, X. Li, and L. Gao, "A New Reinforcement Learning based Learning Rate Scheduler for Convolutional Neural Network in Fault Classification," *IEEE Transactions on Industrial Electronics*, pp. 1– 1, 2020, doi: 10.1109/TIE.2020.3044808.
- [170] A. Moradzadeh and K. Pourhossein, "Location of Disk Space Variations in Transformer Winding using Convolutional Neural Networks," in 2019 54th International Universities Power Engineering Conference, UPEC 2019 - Proceedings, 2019, pp. 1–5, doi: 10.1109/UPEC.2019.8893596.
- [171] M. A. Hannan *et al.*, "SOC Estimation of Li-ion Batteries with Learning Rate-Optimized Deep Fully Convolutional Network," *IEEE Transactions on Power Electronics*, pp. 1–1, 2020, doi: 10.1109/TPEL.2020.3041876.
- [172] W. Kirchgassner, O. Wallscheid, and J. Boecker, "Estimating Electric Motor Temperatures with Deep Residual Machine Learning," *IEEE Transactions on Power Electronics*, pp. 1–1, 2020, doi: 10.1109/TPEL.2020.3045596.
- [173] S. Xie and J. Ren, "Tracking Control Using Recurrent-neuralnetwork-based Inversion Model: a Case Study on a Piezo Actuator," *IEEE Transactions on Industrial Electronics*, pp. 1–1, 2020, doi: 10.1109/TIE.2020.3037989.
- [174] J. Fei and Y. Chen, "Dynamic Terminal Sliding-Mode Control for Single-Phase Active Power Filter Using New Feedback Recurrent Neural Network," *IEEE Transactions on Power Electronics*, vol. 35, no. 9, pp. 9906–9924, Sep. 2020, doi: 10.1109/TPEL.2020.2974470.
- [175] F. Yang, W. Li, C. Li, and Q. Miao, "State-of-charge estimation of lithium-ion batteries based on gated recurrent neural network," *Energy*, vol. 175, pp. 66–75, May 2019, doi: 10.1016/j.energy.2019.03.059.
- [176] A. Shrestha and A. Mahmood, "Review of deep learning algorithms and architectures," *IEEE Access*, vol. 7, pp. 53040–53065, 2019, doi: 10.1109/ACCESS.2019.2912200.
- [177] S. Hochreiter and J. Schmidhuber, "Long Short-Term Memory," *Neural Computation*, vol. 9, no. 8, pp. 1735–1780, Nov. 1997, doi: 10.1162/neco.1997.9.8.1735.
- [178] K. Cho et al., "Learning phrase representations using RNN encoderdecoder for statistical machine translation," in EMNLP 2014 - 2014 Conference on Empirical Methods in Natural Language Processing, Proceedings of the Conference, 2014, pp. 1724–1734, doi: 10.3115/v1/d14-1179.
- [179] X. Fu, S. Li, and I. Jaithwa, "Implement Optimal Vector Control for LCL-Filter-Based Grid-Connected Converters by Using Recurrent Neural Networks," *IEEE Transactions on Industrial Electronics*, vol. 62, no. 7, pp. 4443–4454, Jul. 2015, doi: 10.1109/TIE.2015.2390140.
- [180] S. G. Chen, F. J. Lin, C. H. Liang, and C. H. Liao, "Intelligent Maximum Power Factor Searching Control Using Recurrent Chebyshev Fuzzy Neural Network Current Angle Controller for SynRM Drive System," *IEEE Transactions on Power Electronics*, vol. 36, no. 3, pp. 3496–3511, Mar. 2021, doi: 10.1109/TPEL.2020.3016709.
- [181] S. Hou, Y. Chu, and J. Fei, "Robust Intelligent Control for a Class of Power-electronic Converters Using Neuro-Fuzzy Learning Mechanism," *IEEE Transactions on Power Electronics*, pp. 1–1, 2021, doi: 10.1109/TPEL.2021.3049553.
- [182] D. Li, Z. Zhang, P. Liu, Z. Wang, and L. Zhang, "Battery Fault Diagnosis for Electric Vehicles Based on Voltage Abnormality by Combining the Long Short-Term Memory Neural Network and the Equivalent Circuit Model," *IEEE Transactions on Power Electronics*, vol. 36, no. 2, pp. 1303–1315, Feb. 2021, doi: 10.1109/TPEL.2020.3008194.
- [183] S. Zakeri and M. Geravanchizadeh, "Supervised binaural source separation using auditory attention detection in realistic scenarios," *Applied Acoustics*, vol. 175, p. 107826, Apr. 2021, doi: 10.1016/j.apacoust.2020.107826.
- [184] B. D and K. B.S, "Rider Deep LSTM Network for Hybrid Distance Score-based Fault Prediction in Analog Circuits," *IEEE Transactions* on *Industrial Electronics*, pp. 1–1, 2020, doi: 10.1109/TIE.2020.3028796.
- [185] A. Moradzadeh, S. Zakeri, M. Shoaran, B. Mohammadi-Ivatloo, and F. Mohammadi, "Short-term load forecasting of microgrid via hybrid

support vector regression and long short-term memory algorithms," *Sustainability (Switzerland)*, vol. 12, no. 17, p. 7076, Aug. 2020, doi: 10.3390/su12177076.

- [186] Y. Ma, D. Oslebo, A. Maqsood, and K. Corzine, "Dc Fault Detection and Pulsed Load Monitoring using Wavelet Transform-fed LSTM Autoencoders," *IEEE Journal of Emerging and Selected Topics in Power Electronics*, pp. 1–1, 2020, doi: 10.1109/JESTPE.2020.3019382.
- [187] N. Koroniotis, N. Moustafa, and E. Sitnikova, "Forensics and Deep Learning Mechanisms for Botnets in Internet of Things: A Survey of Challenges and Solutions," *IEEE Access*, vol. 7, pp. 61764–61785, 2019, doi: 10.1109/ACCESS.2019.2916717.
- [188] R. Pahič, Z. Lončarević, A. Gams, and A. Ude, "Robot skill learning in latent space of a deep autoencoder neural network," *Robotics and Autonomous Systems*, vol. 135, p. 103690, Jan. 2021, doi: 10.1016/j.robot.2020.103690.
- [189] S. Ma, M. Chen, J. Wu, Y. Wang, B. Jia, and Y. Jiang, "High-Voltage Circuit Breaker Fault Diagnosis Using a Hybrid Feature Transformation Approach Based on Random Forest and Stacked Autoencoder," *IEEE Transactions on Industrial Electronics*, vol. 66, no. 12, pp. 9777–9788, Dec. 2019, doi: 10.1109/TIE.2018.2879308.
- [190] A. Moradzadeh, K. Pourhossein, B. Mohammadi-Ivatloo, and F. Mohammadi, "Locating Inter-Turn Faults in Transformer Windings Using Isometric Feature Mapping of Frequency Response Traces," *IEEE Transactions on Industrial Informatics*, pp. 1–1, 2020, doi: 10.1109/tii.2020.3016966.
- [191] Y. Lei, F. Jia, J. Lin, S. Xing, and S. X. Ding, "An Intelligent Fault Diagnosis Method Using Unsupervised Feature Learning Towards Mechanical Big Data," *IEEE Transactions on Industrial Electronics*, vol. 63, no. 5, pp. 3137–3147, May 2016, doi: 10.1109/TIE.2016.2519325.
- [192] W. Zhang, D. Xu, P. N. Enjeti, H. Li, J. T. Hawke, and H. S. Krishnamoorthy, "Survey on fault-tolerant techniques for power electronic converters," *IEEE Transactions on Power Electronics*, vol. 29, no. 12, pp. 6319–6331, Dec. 2014, doi: 10.1109/TPEL.2014.2304561.
- [193] S. Cui, J. Hu, and R. De Doncker, "Fault-Tolerant Operation of a TLC-MMC Hybrid DC-DC Converter for Interconnection of MVDC and HVdc Grids," *IEEE Transactions on Power Electronics*, vol. 35, no. 1, pp. 83–93, Jan. 2020, doi: 10.1109/TPEL.2019.2911853.
- [194] L. Sun, Q. T. An, L. Z. Sun, and K. Zhao, "Switching function modelbased fast-diagnostic method of open-switch faults in inverters without sensors," *IEEE Transactions on Power Electronics*, vol. 26, no. 1, pp. 119–126, Jan. 2011, doi: 10.1109/TPEL.2010.2052472.
- [195] J. Li, A. Q. Huang, Z. Liang, and S. Bhattacharya, "Analysis and Design of Active NPC (ANPC) Inverters for Fault-Tolerant Operation of High-Power Electrical Drives," *IEEE Transactions on Power Electronics*, vol. 27, no. 2, pp. 519–533, Feb. 2012, doi: 10.1109/TPEL.2011.2143430.
- [196] Ui-Min Choi, Kyo-Beum Lee, and F. Blaabjerg, "Diagnosis and Tolerant Strategy of an Open-Switch Fault for T-Type Three-Level Inverter Systems," *IEEE Transactions on Industry Applications*, vol. 50, no. 1, pp. 495–508, Jan. 2014, doi: 10.1109/TIA.2013.2269531.
- [197] S.-Z. Xu, C.-J. Wang, and Y. Wang, "An Improved Fault-Tolerant Control Strategy for High-Power ANPC Three-Level Inverter Under Short-Circuit Fault of Power Devices," *IEEE Access*, vol. 7, pp. 55443–55457, 2019, doi: 10.1109/ACCESS.2019.2913594.
- [198] S. Yang, Y. Tang, and P. Wang, "Seamless Fault-Tolerant Operation of a Modular Multilevel Converter with Switch Open-Circuit Fault Diagnosis in a Distributed Control Architecture," *IEEE Transactions* on *Power Electronics*, vol. 33, no. 8, pp. 7058–7070, Aug. 2018, doi: 10.1109/TPEL.2017.2756849.
- [199] F. Bento and A. J. M. Cardoso, "Open-Circuit Fault Diagnosis and Fault Tolerant Operation of Interleaved DC-DC Boost Converters for Homes and Offices," *IEEE Transactions on Industry Applications*, vol. 55, no. 5, pp. 4855–4864, Sep. 2019, doi: 10.1109/TIA.2019.2916825.
- [200] A. Mohammadpour, S. Sadeghi, and L. Parsa, "A Generalized Fault-Tolerant Control Strategy for Five-Phase PM Motor Drives Considering Star, Pentagon, and Pentacle Connections of Stator Windings," *IEEE Transactions on Industrial Electronics*, vol. 61, no. 1, pp. 63–75, Jan. 2014, doi: 10.1109/TIE.2013.2247011.
- [201] H. Guo, S. Guo, J. Xu, and X. Tian, "Power Switch Open-Circuit Fault Diagnosis of Six-Phase Fault Tolerant Permanent Magnet Synchronous Motor System Under Normal and Fault-Tolerant

Operation Conditions Using the Average Current Park's Vector Approach," *IEEE Transactions on Power Electronics*, vol. 36, no. 3, pp. 2641–2660, Mar. 2021, doi: 10.1109/TPEL.2020.3017637.

- [202] M. Shahbazi, P. Poure, S. Saadate, and M. R. Zolghadri, "Faulttolerant five-leg converter topology with FPGA-Based reconfigurable control," *IEEE Transactions on Industrial Electronics*, vol. 60, no. 6, pp. 2284–2294, Jun. 2013, doi: 10.1109/TIE.2012.2191754.
- [203] J. Andreu, I. Kortabarria, E. Ibarra, I. M. de Alegria, and E. Robles, "A new hardware solution for a fault tolerant matrix converter," in 2009 35th Annual Conference of IEEE Industrial Electronics, 2009, pp. 4469–4474, doi: 10.1109/IECON.2009.5414858.
- [204] J. Liu, J. Yao, S. Yang, and K. Wang, "Loss analysis of two kinds of flexible HVDC converters," in *Conference Proceedings - 2012 IEEE 7th International Power Electronics and Motion Control Conference - ECCE Asia, IPEMC 2012*, 2012, vol. 3, pp. 1669–1674, doi: 10.1109/IPEMC.2012.6259086.
- [205] Y. Zuo, X. Zhu, X. Si, and C. H. T. Lee, "Fault-Tolerant Control for Multiple Open-Leg Faults in Open-End Winding Permanent Magnet Synchronous Motor System Based on Winding Reconnection," *IEEE Transactions on Power Electronics*, vol. 36, no. 5, pp. 6068–6078, May 2021, doi: 10.1109/TPEL.2020.3030237.
- [206] J. A. Restrepo, A. Berzoy, A. E. Ginart, J. M. Aller, R. G. Harley, and T. G. Habetler, "Switching Strategies for Fault Tolerant Operation of Single DC-link Dual Converters," *IEEE Transactions on Power Electronics*, vol. 27, no. 2, pp. 509–518, Feb. 2012, doi: 10.1109/TPEL.2011.2161639.
- [207] S. Ceballos, J. Pou, J. Zaragoza, E. Robles, J. L. Villate, and J. L. Martín, "Fault-tolerant neutral-point-clamped converter solutions based on including a fourth resonant leg," *IEEE Transactions on Industrial Electronics*, vol. 58, no. 6, pp. 2293–2303, Jun. 2011, doi: 10.1109/TIE.2010.2069075.
- [208] X. Zhou, J. Sun, H. Li, M. Lu, and F. Zeng, "PMSM Open-Phase Fault-Tolerant Control Strategy Based on Four-Leg Inverter," *IEEE Transactions on Power Electronics*, vol. 35, no. 3, pp. 2799–2808, Mar. 2020, doi: 10.1109/TPEL.2019.2925823.
- [209] S. Ceballos et al., "Three-Level Converter Topologies With Switch Breakdown Fault-Tolerance Capability," *IEEE Transactions on Industrial Electronics*, vol. 55, no. 3, pp. 982–995, Mar. 2008, doi: 10.1109/TIE.2008.917072.
- [210] Z. Yi, S. Hongge, and X. Bin, "Optimization of neutral shift in cellfault treatment for cascaded H-bridge inverter," in *Proceedings of the* 11th International Conference on Electrical Machines and Systems, ICEMS 2008, 2008, pp. 1683–1685.
- [211] S. Farzamkia, H. Iman-Eini, M. Noushak, and A. Hadizadeh, "Improved Fault-Tolerant Method for Modular Multilevel Converters by Combined DC and Neutral-Shift Strategy," *IEEE Transactions on Industrial Electronics*, vol. 66, no. 3, pp. 2454–2462, Mar. 2019, doi: 10.1109/TIE.2018.2823659.
- [212] S. M. Kim, J. S. Lee, and K. B. Lee, "A Modified Level-Shifted PWM Strategy for Fault-Tolerant Cascaded Multilevel Inverters with Improved Power Distribution," *IEEE Transactions on Industrial Electronics*, vol. 63, no. 11, pp. 7264–7274, 2016, doi: 10.1109/TIE.2016.2547917.
- [213] P. W. Hammond, "Enhancing the reliability of modular mediumvoltage drives," *IEEE Transactions on Industrial Electronics*, vol. 49, no. 5, pp. 948–954, Oct. 2002, doi: 10.1109/TIE.2002.803172.
- [214] F. Carnielutti, H. Pinheiro, and C. Rech, "Generalized carrier-based modulation strategy for cascaded multilevel converters operating under fault conditions," *IEEE Transactions on Industrial Electronics*, vol. 59, no. 2, pp. 679–689, Feb. 2012, doi: 10.1109/TIE.2011.2157289.
- [215] W. Song and A. Q. Huang, "Fault-tolerant design and control strategy for cascaded H-bridge multilevel converter-based STATCOM," *IEEE Transactions on Industrial Electronics*, vol. 57, no. 8, pp. 2700–2708, Aug. 2010, doi: 10.1109/TIE.2009.2036019.
- [216] Gum Tae Son et al., "Design and Control of a Modular Multilevel HVDC Converter With Redundant Power Modules for Noninterruptible Energy Transfer," *IEEE Transactions on Power Delivery*, vol. 27, no. 3, pp. 1611–1619, Jul. 2012, doi: 10.1109/TPWRD.2012.2190530.
- [217] F. H. Khan and L. M. Tolbert, "Multiple-Load–Source Integration in a Multilevel Modular Capacitor-Clamped DC–DC Converter Featuring Fault Tolerant Capability," *IEEE Transactions on Power Electronics*, vol. 24, no. 1, pp. 14–24, Jan. 2009, doi:

10.1109/TPEL.2008.2006055.

[218] K. Shen et al., "A modulation reconfiguration based fault-tolerant control scheme for modular multilevel converters," in Conference Proceedings - IEEE Applied Power Electronics Conference and Exposition - APEC, 2013, pp. 3251–3255, doi: 10.1109/APEC.2013.6520766.



Arash Moradzadeh (S'19) was born in Tabriz, Iran, in 1991. He received the B.S. degree in electrical power engineering and the M.S. degree in power electronics and electrical machines from Islamic Azad University of Tabriz, Tabriz, Iran, respectively in 2016 and 2019. He is currently working toward the Ph.D. degree in power electrical engineering

with the University of Tabriz, Tabriz, Iran.

His current research interests include power and energy systems, cyber-physical systems, energy management, intelligent energy systems, transient in power systems, diagnostics and condition monitoring of power transformers, frequency response analysis, and application of data mining methods to design, optimize and analysis of energy systems.



Behnam Mohammadi-Ivatloo (S'09– M'12–SM'17) received the B.Sc. degree in electrical engineering from University of Tabriz, Tabriz, Iran, in 2006, and the M.Sc. and Ph.D. degrees both in electrical engineering from Sharif University of Technology, Tehran, Iran, in 2008, all with honors. He is currently a Professor with Faculty of Electrical and Computer

Engineering, University of Tabriz, Tabriz, Iran. His main area of research is economics, operation, and planning of intelligent energy systems in a competitive market environment. He was editor of several books published by Springer and Elsevier. He as an Associate Editor of the IEEE TRANSACTIONS ON POWER SYSTEMS and IEEE ACCESS journals.



Kazem Pourhossein received the B.Sc. degree in electrical engineering from K. N. Toosi University of technology, Tehran, Iran, in 2002, the M.Sc. degree in electric power engineering from Zanjan University, Zanjan, Iran, in 2006, and the Ph.D. degree from Science & Research branch of Azad University in 2012, Tehran, Iran. He was with R&D

department of IranTransfo from 2004 to 2006 working on transient phenomena in power transformers. He is currently an assistant professor in Tabriz branch of Azad University. His research interests include power transformer diagnosis, condition monitoring of power apparatus, high voltage, transients in power systems, renewable and sustainable energy systems and pattern recognition of energy systems.



Amjad Anvari-Moghaddam (S'10–M'14–SM'17) received the Ph.D. degree (Hons.) from University of Tehran in 2015 in Power Systems Engineering. From 2015 until 2019 he was a Postdoctoral Research Fellow at Aalborg University. Currently, he is an Associate Professor and Leader of Intelligent Energy Systems and

Flexible Markets (iGRIDS) Research Group at the Department of Energy (AAU Energy), Aalborg University where he is also acting as the Vice-Leader of Power Electronic Control, Reliability and System Optimization (PESYS) and the coordinator of Integrated Energy Systems Laboratory (IES-Lab). His research interests include planning, control and operation management of microgrids, renewable/hybrid power systems and integrated energy systems with appropriate market mechanisms. He has (co)authored more than 250 technical articles, five books and nine book chapters in the field. Dr. Anvari-Moghaddam serves as the Associate Editor of the IEEE TRANSACTIONS ON POWER SYSTEMS, IEEE ACCESS, IEEE SYSTEMS JOURNAL, IEEE OPEN ACCESS JOURNAL OF POWER AND ENERGY, and IEEE POWER ENGINEERING LETTERS. He is the Vice-Chair of IEEE-PES Danish Chapter and serves as a Technical Committee Member of several IEEE PES/IES/PELS and CIGRE working groups. He was the recipient of 2020 DUO - India Fellowship Award, DANIDA Research Fellowship grant from the Ministry of Foreign Affairs of Denmark in 2018, IEEE-CS Outstanding Leadership Award 2018 (Halifax, Nova Scotia, Canada), and the 2017 IEEE-CS Outstanding Service Award (Exeter-UK).